PROSPECTIVE EARLY CHILDHOOD TEACHERS' CONCEPTIONS OF SCIENCE INSTRUCTION

Supreeya Muimongkol

Dissertation Prepared for the Degree of

DOCTOR OF PHILOSOPHY

UNIVERSITY OF NORTH TEXAS

August 2019

APPROVED:

Dina C. Castro, Major Professor Karthigeyan Subramaniam, Committee Co-Chair Carol D. Wickstrom, Committee Member Tran Nguyen Templeton, Committee Member Jim Laney, Chair of the Department of Teacher Education and Administration Randy Bomer, Dean of the College of Education Victor Prybutok, Dean of the Toulouse Graduate School Muimongkol, Supreeya. *Prospective Early Childhood Teachers' Conceptions of Science Instruction*. Doctor of Philosophy (Curriculum and Instruction), August 2019, 148 pp., 24 tables, 25 figures, 4 appendices, references, 183 titles.

The purpose of this study was to investigate prospective teachers' conceptions of science instruction prior to the commencement of their early childhood education methods courses. A sample of 100 prospective teachers' drawings and narratives were analyzed using the STLP3 instrument and inductive analysis respectively. Merging findings from the aforementioned analyses resulted in 10 distinct conceptions and these included: (a) science instruction (1) as a demonstration, (2) as engagement, (3) as doing experiments, (4) as inquiry, and (5) as observation all underscored by active learning and social, implementation and cognitive dimensions; (b) science instruction (6) as implementing safety, and (7) as an interaction both underscored by active learning and a social dimension; and (c) science instruction (8) as a demonstration, and (9) as engagement both underscored by passive learning and an affective dimension; and (d) science instruction (10) as enjoyment underscored by social and affective dimensions. These findings reflected the complexity and multidimensionality of the prospective teachers' conceptions of science instruction. Implications include the need for teacher educators to situate their prospective teachers' prior and new knowledge of early childhood science instruction within theoretical frames rather than simply relying on prospective teachers' knowledge of science instruction from K-12 experiences. Implications for future research include the need to study prospective teachers' conceptions prior to and at the end of early childhood methods courses.

Copyright 2019

by

Supreeya Muimongkol

ACKNOWLEDGEMENTS

This dissertation would not have been possible without the guidance and encouragement that supported my work over the last five years at the University of North Texas. I would like to express my sincere gratitude to those who have made it possible for me to travel along the journey of a Doctoral degree in Curriculum & Instruction.

To my exceptional Committee: I give my deepest thanks to Dr. Dina C. Castro, my major professor, for her guidance in helping me achieve success with the dissertation. Special thanks to Dr. Karthigeyan Subramaniam, my committee co-chair for providing endless support, for his patience and time in guiding me through the methodology, and for teaching me how to think and write as a researcher. Special thanks also go to Dr. Carol D. Wickstrom and Dr. Tran Nguyen Templeton for the insights and thoughtful questions, they shared while serving as dissertation committee members.

Thanks to Dr. Carol Hagen, for her guidance regarding my findings and discussion; without her support, my dissertation could not have been completed. To my sister, Dr. Chaowanee Kumlertluk Moore, for her suggestions on how to write a dissertation as a professional.

To my family, especially my father and mother, for being the strongest and most loving persons I know. Without their support, I could not have reached my study goal. Thanks to my husband, Nu, for being patient along the course of my study in the U.S. and to my children, Khai Mook (Pearl) and Khai Tun (Gracie), the greatest loves of my life, for being my motivation, strength, and the embodiment of beauty and joy.

iii

TABLE OF CONTENTS

ACKNOWLEDGEMENTSiii		
LIST OF TABLESvii		
LIST OF FIGURESix		
CHAPTER 1. INTRODUCTION		
Statement of the Problem5		
Frameworks6		
Theoretical Framework of Sociocultural Theory		
Teachers' Conceptions of Teaching7		
Purpose of the Study8		
Research Questions8		
Significance of the Study8		
Research Method9		
Operational Definitions9		
Summary 11		
CHAPTER 2. RELATED LITERATURE		
Early Childhood Science Education		
Science Content Standards for Early Childhood Education		
Effective Early Childhood Science Instruction		
Prospective Early Childhood Teachers' Conceptions of Science Education		
The Role of Drawing in Educational Research25		
The Science Teaching and Learning Portrayals of Professional Practices (STLP3)		
The Sociocultural Perspective		
The Sociocultural Perspective on Science Education		
Summary		
CHAPTER 3. METHODOLOGY		
Research Design		

Participants	40
Research Context	
Data Collection and Data Sources	
Drawing Procedures	
Narratives	
Data Analysis	
Analysis of Drawings	45
Analysis of Narratives	
Trustworthiness	50
Summary	
	F 2
CHAPTER 4. FINDINGS	
Participants Conceptions of Science Instruction: Drawings	
Participants' Conceptions of Science Instruction: Narratives	
Demonstrations	
Engagement	
Experiment	
Safety	85
Inquiry	88
Observations	91
Enjoyment	93
Interaction	95
Synthesis of Findings from Drawings and Narratives	
Summary	
CHAPTER 5. DISCUSSION	
Findings in Relation to Standards	
Findings in Relation to Current Instructional Strategies: Pre-K to 3 rd Grade So Instruction	cience
Findings in Relation to the Sociocultural Theory	108
Limitations	
Educational Implications	

Implications for the Early Childhood Teacher Educator
Implications for the Early Childhood Teacher Preparation Programs
Implications for Future Research112
Conclusion
APPENDIX A. IRB
APPENDIX B. SCIENCE TEACHING AND LEARNING PORTRAYALS OF PROFESSIONAL PRACTICE (STLP3) SCORING RUBRIC
APPENDIX C. SCORING RUBRIC SUPPLEMENTAL INFORMATION SHEET
APPENDIX D. EXAMPLES OF PARTICIPANTS' DRAWINGS 122
REFERENCES

LIST OF TABLES

Page

Table 1. Number of Participants by Race and Ethnicity
Table 2. The Qualitative Criteria for Assessing Research Quality and Rigor 50
Table 3. Scores and Percentages of Participants' Conceptions of Science Instruction byDimensions Present in the STLP3 Rubric (N = 100)54
Table 4. Numbers and Percentages of Participants' Conceptions Science Instruction asDemonstrations for Passive Learning and for Active Learning
Table 5. Examples of Participants' Conceptions of Science Instruction as Demonstrations forPassive Learning
Table 6. Examples of Participants' Science Instruction as Demonstrations for Active Learning . 74
Table 7. Numbers and Percentages of Participants' Conceptions of Science Instruction as Engagement 77
Table 8. Excerpts from Participants' Narratives that were Predominantly Associated with theConception of Science Instruction as Engagement
Table 9. Number of Participants' Narrative Conceptions of Science Instruction as Experiment(involving Active Learning)
Table 10. Participants' Narratives Indicating Conceptions of Science Instruction as Experiment(in Active Learning)83
Table 11. Number of Participants' Narrative Conceptions of Science Instruction as ImplementingScience Safety Guidelines (Active Learning)86
Table 12. Participants' Narratives Indicating Conceptions of Science Instruction as ImplementingScience Safety Guidelines (in Active Learning)86
Table 13. Number of Participants' Narrative Conceptions of Science Instruction as Being Inquiry-Based (Inquiry) (in Active Learning)
Table 14. Excerpts from Participants' Narratives Conceptions of Science Instruction as Being Inquiry Based (Inquiry) 89
Table 15. Number of Participants' Narrative Conceptions of Science Instruction as StudentsObserving Scientific Phenomena (Observations)91

Table 16. Excerpts from Participants' Narratives Conceptions of Science Instruction as Observations 92
Table 17. Number of Participants' Narrative Conceptions of Science Instruction as Enjoyment.94
Table 18. Excerpts from Participants' Narratives Conceptions of Science Instruction as Observations 94
Table 19. Number of Participants' Narrative Conceptions of Science Instruction as Interaction 95
Table 20. Excerpts from Participants' Narratives Conceptions of Science Instruction as Interaction
Table 21. Identifying the Numbers of Percentage of Participants' Conceptions of Science Instruction from Participants' Drawings, Participants' Narratives, and Science Content
Table 22. Examples of Merging Themes of Participants' Conceptions of Science Instruction for Children in Pre-K to 3 rd Grade Classroom
Table 23. Comparing the Participants' Narratives Conceptions of Science Instruction for Children in Pre-K to 3 rd Grade between Active Learning and Passive Learning
Table 24. Major Themes Resulting from Merging of Findings from Drawings and Narratives 103

LIST OF FIGURES

Figure 1. Science content standards for early childhood education (NRC, 1996; Holweg &Hill,2003).16
Figure 2. A diagrammatic representation of researcher's articulation of the sociocultural theory
Figure 3. The purpose of methodology 40
Figure 4. Diagram indicating the number of participants by race and ethnicity
Figure 5. Example of participants' course background in a teacher preparation program 43
Figure 6. Data analysis framework
Figure 7. Steps to ensure trustworthiness
Figure 8. A participant's drawing receiving a score of 4 for science instruction as a social activity
Figure 9. A participant's drawing receiving a score of 0 for science instruction as a social activity.
Figure 10. A participant's drawing receiving a score of 4 for science instruction as an affective endeavor
Figure 11. A participant's drawing receiving a score of 0 for science instruction as an affective endeavor
Figure 12. A participant's drawing receiving a score of 4 for science instruction as an implementation of activities for learning
Figure 13. A participant's drawing receiving a score of 0 for science instruction as an implementation of activities for learning
Figure 14. A participant's drawing receiving a score of 4 for science instruction as a cognitive activity
Figure 15. A participant's drawing has a score of 0 for science instruction as a cognitive activity
Figure 16. An example of the highest score of a participant's drawing of conceptions of science instruction

Figure 17. The diagram indicates the percentages of participants' narrative conception of science instruction as a type of demonstration (passive learning)
Figure 18. The diagram indicates number of percentages of participants' narrative conception of science instruction as a type of demonstration (active learning)
Figure 19. Diagram of the percentages of participants' narrative conceptions science instruction as engagement (active learning)
Figure 20. Diagram of the percentages of participants' narrative conceptions of science instruction as experiment (in active learning)
Figure 21. Diagram of the percentages of participants' narrative demonstrating conceptions of science instruction as implementing science safety guidelines (in active learning)
Figure 22. Diagram of the percentages of participants' narrative conceptions of science instruction as being inquiry-based (Inquiry) (in active learning)
Figure 23. Diagram of the percentages of participants' narrative conceptions of science instruction as students observing scientific phenomena (observations) (in active learning) 92
Figure 24. Diagram of the percentages of participants' narrative conceptions of science instruction as enjoyment (passive learning)
Figure 25. Diagram of the percentages of participants' narrative conceptions of science instruction as interaction (in active learning)

CHAPTER 1

INTRODUCTION

The field of early childhood education (ECE) has recently increased its emphasis on the learning of specific subjects and science is one of these (Areljung, Ottander & Due, 2017; The Organisation for Economic Co-operation and Development (OECD), 2006). Science is an important subject for children because it is linked to their lives and environment; for instance, children can learn about the organs of the body and their functions. Children become more engaged in science when they can use their senses or engage in hands-on activities because these allow them to be more active in the learning process. Science instruction is becoming increasingly important in preschool through 3rd grade classrooms; therefore, it has become necessary for teacher educators to advocate for conceptualized science instruction for prospective early childhood teachers.

Teachers across the country are working hard to equip children with the skills needed to prepare for science, technology, engineering, and mathematics (STEM) careers to be well equipped as part of the modern workforce, and an educated society knowledgeable about technology is recognized as a need for the 21st century. Improving the quality of today's teachers is a crucial part of the effort to harmonize education with the development of society and technology.

The National Research Council (NRC, 2012) directs teachers toward careers in highquality science teaching with clear criteria to indicate what teachers can do to support science learners at all age levels. Therefore, the NRC calls for teachers to practice active involvement in scientific investigation; it also introduces them to resources to expand their knowledge and

ability, to build their scientific understanding, abilities, and attitudes, and to make use of their collaborative science learning experiences in the classroom.

Early childhood and science education research supports the idea that science should be introduced to young children as early as possible (Pendergast, Lieberman-Betz, & Vall, 2017). The National Association for the Education of Young Children (NAEYC) and the National Association of Early Childhood Specialist in State Departments of Education (NAECS/SDE) have raised critical questions regarding early childhood education, including "What should children be taught in the years from birth through age eight? How would we know if they are developing well and learning what we want them to learn? In addition, how could we decide whether programs for children from infancy through the primary grades are doing a good job?" (NAEYC & NAECS/SDE, 2003).

The aim of early childhood science education is for young children to find out about their everyday surroundings (Brenneman, 2011; NSTA, 2014), especially since young children have intrinsic motivation to engage in science and science learning (French, 2004). The National Science Teachers Association (NSTA) affirms that learning science in their early years can foster children's curiosity and enjoyment in exploring the world around them and lay the foundation for a progression of science learning in pre-K to 12 settings and throughout their lives (NSTA, 2014). Teaching science early, during preschool, is important (Allen, 2017; Ashbrook, 2018; Clements, 2016; Inan & Inan, 2015; IPST, 2011; Kim & Han, 2015; NGSS Lead States, 2013; NSTA, 2014; Trundle, 2009; Weidong & Bird, 2016). Thus, not only does early childhood science education support children's better understanding of science concepts, acquisition of scientific thinking skills, and improvement in various developmental domains, but also the whole of

children's development is furthered with the help of both indoor and outdoor learning environments (e.g., nature-related activities) (Sackes, Trundle, & Flevares, 2009).

Several policy directives are related to early childhood science education: they include (a) establishing a system of learning networks; (b) providing and promoting the prenatal care, child-rearing practices, and education necessary for child development; (c) preparing every child for entry into primary school; (d) reforming teacher pre- and in-service education; (e) improving curriculum content and teaching-learning processes at all levels and in all types of education; and (f) promoting and supporting the participation of families, local communities, social institutions, and mass media in the educational process (IPST, 2011;The MOE, 2003; UNESCO, 2006).

Early childhood educators play a key role in offering the first formal learning environments to children, in which they can gain science experiences. In addition, preparation and professional development programs to improve teacher quality are needed to promote and improve science literacy in the classroom. The National Science Teachers Association (NSTA, 2017) supports teacher preparation aligned with the goals and guidance provided by *A Framework for K–12 Science Education* (NRC, 2012), *Next Generation Science Standards* (NGSS Lead States, 2013), and *Science Teachers' Learning* (NRC, 2015). The NSTA has adopted and applied the *Standards for Science Teacher Preparation* (NSTA, 2012). It is committed to increasing the number of highly qualified science teachers by ensuring that all those entering the profession demonstrate a deep understanding of pure and applied science and have the knowledge and skills required to teach students science in age-appropriate, meaningful ways (NSTA, 2017).

In the United States, science, technology, engineering and mathematics (STEM) topics have been increasingly recognized as appropriate and relevant for children to explore in early childhood settings (California Department of Education, 2012; Greenfield et al., 2009; McClure et al., 2017; National Science Foundation, 2013). Teaching STEM skills involves developing scenarios that reflect the kinds of science content to which preschool-age children are likely to be introduced in their classroom experiences. Despite the emphasis on reform efforts to improve science teacher preparation and professional development, particularly those who teach at the early childhood level, many teachers report entering the classroom feeling inadequately prepared to teach science (Kelly, 2000; Murphy, Neil, & Beggs, 2007). Teachers are limited in their content knowledge and self- efficacy for science teaching (Sackes, Trundle, Bell, & O'Connell, 2011; Tu, 2006).

Consequently, investigating prospective teachers' conceptions of science education is important because teachers' conceptions in both informal and formal settings can reasonably be expected to affect teaching practices (Subramaniam, 2013). Specifically, preparing teachers to teach science effectively in early childhood classrooms is an area of research that is becoming more important for both prospective teachers and faculty members. Researchers have long been aware of the popularity of prospective teachers' conceptions of teaching and learning. Teacher educators need to realize that the teaching process is becoming an important variable in the development of students' epistemological beliefs and a crucial factor for opening doors to new vistas and stimulating novel alternative solutions and ideas in students. Similarly, teacher educators may benefit by helping teachers to explore their teaching and learning conceptions regarding the students in their classes (Mahasneh, 2018).

Statement of the Problem

In recent years, a growing number of researchers in teacher education have focused on the development and design of efforts to prepare prospective primary education teachers (Brown & Englehardt ,2017; Eckhoff, 2017; Mckinnon & Perara, 2015; Subramaniam, 2013; Subramaniam et al., 2018). However, there is a particular limitation on prospective early childhood teachers. Science methods courses taken by prospective early childhood teachers influence their future instruction in early childhood classrooms (Seung-Yoeun, 2010). There are a number of factors found to influence early childhood teachers' teaching of science in the classroom. Sackes (2014) identifies factors that influence early childhood teachers' decisions to devote less time to teaching science in their classroom as follows: (a) their limited science and pedagogical content knowledge , (b) pressure to teach content areas other than science (Greenfield et al., 2009; Nayfeld et al., 2013), (c) limited availability of science-related materials (Greenfield et al., 2009), (d) perceptions that young children are incapable of learning science concepts (Fleer, 2009), and (e) low teacher self-efficacy for teaching science (Maier, Greenfield, & Bulotsky-Shearer, 2013; Nayfeld et al., 2013).

In several countries, including Sweden, Belgium, New Zealand, Italy, and Australia, and in some states in the US, science is being taught to students at young ages. According to Andersson and Gullberg (2014), effective teaching of science in early childhood requires a set of competencies needed by early childhood educators to conduct classroom science activities. These competencies include teaching early childhood science following theoretical frameworks and using effective curricular approaches. Based on the current literature, it is important to begin the teaching of science in the early years as a required subject area. Effective research-

based teaching strategies must be used so that teachers can help to create a stronger foundation in science learning for pre-K to 3rd grade children. Teaching is a very obscure and complicated process, especially the teaching of science for early childhood. Understanding children is every early childhood teacher's challenge.

Prospective early childhood teachers' conceptions of science teaching are an important determinant of how these teachers carry out instruction in their classrooms. Therefore, it is crucial for prospective teachers to equip themselves with the necessary skills to teach science in an effective way before they graduate. Blevin (2018) emphasizes that the way educators teach science strongly correlates with the way they learned to teach science in their undergraduate teaching preparation courses. In the current literature, there is a strong emphasis on the preparation of prospective early childhood teachers, especially in science instruction.

Frameworks

Theoretical Framework of Sociocultural Theory

The theoretical framework that guides this study is that of sociocultural theory. Sociocultural theory has become increasingly important in discussions associated with early childhood education and curriculum at a theoretical level since the early 1990s (Edwards, 2006). This theory emphasizes the ideas of learning connected to the social and cultural nature of development as advocated by Vygotsky (1987). His work reflects sociocultural influence on the construction of knowledge. Vygotsky's view of how children's mental abilities develop focused on the role of the child's social and cultural world. Vygotsky believed that children depend on others to develop their cognitive skills and abilities. The social organization may be the classroom or culture at large. It is the agent for change for the individual. The social context

and the language discourse are paramount to human learning. Learners develop knowledge as a social activity within the context of institutional and cultural frameworks. The language and culture of the social group plays a crucial role in helping students develop ideas and knowledge.

Teachers' Conceptions of Teaching

According to Subramaniam (2013), there are three types of views or conceptions of teaching: (a) conceptions of teaching as beliefs; (b) conceptions of teaching as belief-driven, thereby influencing and interacting with conceptions of teaching by determining whether the conceptions of teaching are to be selected, stored, or discarded; and (c) conceptions of teaching as different models of learning, which are classified as dimensions, orientations, and complex sets of propositions.

In this study, prospective early childhood teachers' conceptions of teaching science are understood as the specific set of instructional ideas that collectively function as an organizing framework by which prospective early childhood teachers develop their knowledge of science teaching in pre-K to 3rd classrooms. These instructional ideas, derived from K-12 teaching experiences, life experiences, and social, cultural, and linguistic backgrounds, and expressed in actions like decision-making, lesson planning, and teaching, are focused on control of science content, directionality of science teaching, understanding of students' existing science conceptions, expected outcomes of science teaching, and students' utilization of science knowledge. Moreover, the actions collectively function to express the complex sets of propositions that in turn indicate the different orientations or dimensions to which the conceptions of teaching science belong in line with this understanding of the optimal conceptions of teaching. In this study, the instrument called the Science Teaching and Learning

Portrayals of Professional Practices (STLP3) (Mckinnon & Perara, 2015), adapted from Project Nexus (2011), was used to analyze prospective early childhood teachers' conceptions of teaching science from their drawings, and a qualitative analysis approach was used to analyze their conceptions of teaching science from their narratives.

Purpose of the Study

The purpose of the current study is to investigate the prospective early childhood teachers' conceptions of science instruction in an early childhood classroom.

Research Questions

The primary research question for the present study is: What are prospective early childhood teachers' conceptions of science instruction in the early childhood classroom?

Significance of the Study

The significance of the study is twofold. First, this study is significant because there is a need to understand prospective early childhood teachers' prior conceptions of science instruction knowledge pedagogies before they enter a teacher preparation program and before they enroll in a science method course. This is important because there is a limited knowledge base on prospective early childhood teachers' conceptions of teaching science and on how their conceptions of teaching science influence how they learn to teach science. Second, the study is unique because the Science Teaching and Learning Portrayals of Professional Practices (STLP3) (Mckinnon & Perara, 2015), adapted from Project Nexus (2011), has not been used in a formal setting to analyze drawings in research on prospective early childhood teachers and their conceptions of science instruction.

Research Method

The present study uses a qualitative approach to conduct a secondary analysis of data. These data, in the form of drawings and narratives, relate to prospective early childhood teachers' conceptions of science instruction. The analysis of data employs the Science Teaching and Learning Portrayals of Professional Practices (STLP3) instrument (Mckinnon & Perara, 2015), adapted from the Project Nexus (2011), with prospective teachers' drawings, as well as qualitative analysis to analyze prospective teachers' narrative writing. Both drawings and narratives of prospective early childhood teachers were analyzed to identify their conceptions of science instruction for children in pre-K to 3rd grade.

Operational Definitions

The following terms are associated with this study. Terms are defined here to clarify the specific meanings of terminology used in this dissertation.

• *Conceptions of science instruction:* The current study views conceptions of science as *affective*—students should experience excitement, interest and motivation to learn about phenomena in the natural and physical world; as *cognitive*—students will become able to generate, understand, remember, and use concepts, explanations, arguments, models, and facts related to science; as *implementation*—students must manipulate, test, explore, predict, question, observe, and make sense of the natural and physical world; and as *social*—students will participate in scientific activities and learning practice with others, using scientific language and scientific tools (Mckinnon & Perara, 2015).

• *Inquiry*. Inquiry is an intricate cycle of scientific exploration involving the use of observations, asking and answering questions, studying existing knowledge using other media,

investigating, applying experimental evidence to existing schema to construct or refine knowledge, gathering and analyzing data using various tools, developing predictions and explanations of phenomena, and communicating results clearly. Inquiry requires the development of skills to assist in identification of assumptions, use of critical thinking and problem-solving, and consideration of unconventional or differing explanations (NRC, 1996, p. 23).

• Inquiry-based Instruction or Teaching: This approach involves the use of scientific methods (inquiry) in classroom science instruction (National Research Council, 1996). Inquiry-based instruction is primarily focused on "important practices, such as modeling, developing explanations, and engaging in critique and evaluation..." (NRC, 2012, p. 44).*Prospective Early Childhood Teachers:* These are college undergraduate students enrolled in a teacher preparation program preparing to become pre-K to 3rd grade teachers. Participants in this research study will come from this group.

• Science Content Standards: The science content standards outline what students should know, understand, and be able to do in the natural sciences over the course of their K-12 education (NRC, 1996, p. 103).

• *Science Teaching Standards:* The science teaching standards describe what teachers of science at all grade levels should know and be able to do (NRC, 1996, p.27).

• *Teacher preparation (for EC-6 teachers):* This concept is understood as a list of professional teaching competencies to be assessed or a course list to complete, along with development of professional knowledge, skills, and dispositions in a community of learners,

applying reading, observation, and field experiences through group projects and interactions with others (NAEYC Professional Preparation Standards, 2012, p. 23).

• *Teachers' conceptions of teaching and learning:* These are "the beliefs held by teachers about their preferred ways of teaching and learning" (Chan & Elliott, 2004, p. 819), including what teaching and learning mean, as well as the relationship between the teacher and students Mahasneh (2018).

Summary

This first chapter has presented an overview of the study including the importance of early childhood science instruction, sociocultural theory, and teachers' conceptions of teaching, seen as the frameworks of this study. It has clarified the purpose and presented the research questions of the study, which lead to the significance of this study. Next, Chapter 2 is a review of the literature related to science instruction in early childhood classrooms and the theoretical underpinnings that will serve as the framework for the study. Chapter 3 discusses the methodology of the study, including the data collection techniques used to collect the secondary data and the instrument used to analyze the drawings and narratives. Chapter 4 presents the findings from applying the Science Teaching and Learning Portrayals of Professional Practices (STLP3) to analyze prospective early childhood teachers' drawings. An analysis of data from prospective teachers' narratives is presented as an investigation of prospective teachers' conceptions of science instruction. Chapter 5 includes a discussion of the findings in relation to the overall research question and the purpose of the study. Chapter 5 further discusses the implications of examining prospective early childhood teachers' drawings and narratives as it pertains to their present and future conceptions of science construction for

teaching children in pre-K to 3rd grade classrooms, and the resulting insights are also presented in this chapter.

CHAPTER 2

RELATED LITERATURE

This review of literature guides the reader through several topics relevant to the formation of the present study. In order to orient the reader to the theoretical underpinnings of the study, the literature review relates to the theoretical constructs concerning how prospective early childhood teachers conceptualize science instruction. This chapter is divided into six sections as follows.

Section 1 discusses the standards that inform the content and practice of early childhood science education. Section 2 presents the current instructional strategies being utilized in early childhood science education. Section 3 presents the literature on prospective early childhood teachers' conceptions of science education. Section 4 discusses the role of drawings as a method to investigate prospective early childhood teachers' conceptions of teaching science. Section 5 presents the instrument that will be used to analyze the data collected for this study. Section 6 presents the sociocultural perspective that will be used as the theoretical framework to guide and frame the study. The review of literature concludes with a discussion of the scholarly and practical significance of the present study.

Early Childhood Science Education

Most U.S. states have developed and published at least one set of early learning standards or learning expectations that describe what children should learn and be able to do as a result of being in preschool (Brenneman et al., 2009). These standards provide (a) a comprehensive description of the knowledge and skills children should have, (b) a guide for administrators and teachers in designing or choosing curricular experiences for pre-K to 3rd

grade students, and (c) benchmarks for early childhood teachers to assess the quality of their offerings (Buchter et al., 2017; NSTA, 2014). The National Science Teachers Association (NSTA, 2014) also identifies key principles to guide the learning of science among young children; these are the following:

- 1. Children have the capacity to engage in scientific activity and develop an understanding at a conceptual level.
- 2. Adults play a central and important role in helping young children learn science.
- 3. Young children need multiple and varied opportunities to engage in science exploration and discovery.
- 4. Young children develop science skills and knowledge in both formal and informal settings.
- 5. Young children develop science skills and knowledge over time.
- 6. Young children develop science skills and learning by engaging in experiential learning.

These principles emphasize the importance of individual, cultural, and linguistically

responsive learning experiences and environments for young children. They also emphasize that parents and other caregivers can nurture children's natural curiosity about the world around them, creating a positive and safe environment at home for them for exploration and discovery. Additionally, the National Research Council advocates that for formal and Informal learning of science, teachers should use the guideline of the science teaching standard below:

- 1. The planning of inquiry-based science programs
- 2. The actions taken to guide and facilitate student learning
- 3. The assessments made of teaching and student learning
- 4. The development of environments that enable students to learn science
- 5. The creation of communities of science learners

6. The planning and development of the school science program(National Research Council (NRC, 2012)

The heart of science education is effective teaching. Good teachers of science create environments in which they and their students work together as active learners. Teachers use assessments of students and of their own teaching continually to plan and conduct their teaching. They build strong relationships with students and include their students as active members of science learning communities. For this to happen, teachers need support from the rest of the educational system.

Science Content Standards for Early Childhood Education

The National Science Education Standards produced by the National Research Council (NRC, 1996) provides guidelines for the science content and process skills that are appropriate for children of different ages; the document has been used by most of the states in the United States to develop their state science education standards. In addition, the National Science Education Standards have been used as a guide for creating the science education framework in each area. However, most efforts to develop science standards have focused on kindergarten through 12th grade. According to the NRC, science content standards statements include *Science as Inquiry Standards, Physical Science Standards, Life Science Standards, Earth and Space Science Standards, Science and Technology Standards, Science in Personal and Social Perspectives and History and Nature of Science Standards, as shown in Figure 1.*





Unfortunately, according to Trundle & Sackes (2015, pp. 241-242), a survey of the science content standards used by all U.S. states found that only 17 states had science content standards for preschool-kindergarten. Further, only 12 states had separate academic science content standards specifically for preschool (Sackes, Trundle, & Flevares, 2009). A review of the preschool science content standards of those 12 states revealed that the three most common content areas across the states were physical science, earth and space science, and life science. These content areas are derived from the National Science Education Standards for Grades K-

12. Within each common content area, several common themes for preschool were also identified across the states. Seventeen states had science content standards for preschoolkindergarten which identified three science content areas: physical science, earth and space science, and life science.

There were six common themes for physical science:

- 1. Physical properties of objects and materials (e.g., solid-liquid and hard-soft) was the most common theme (found in the standards used by all twelve states)
- 2. Classification of objects and materials based on qualities such as weight and shape (nine states)
- 3. Movement of objects (six states)
- 4. Sound
- 5. Light
- 6. Physical changes

There were four common themes for Earth and Space Science:

- 1. Weather (eight states)
- 2. Day and night (four states)
- 3. Earth materials (three states)
- 4. Seasons (three states)

There were five common themes for Life Science:

- 1. Life cycles of plants and animals (nine states)
- 2. Plant and animal habitats (eight states)
- 3. Classification of plants and animals (seven states)
- 4. Common needs of plants and animals (six states)
- a. Heredity (five states) (Trundle & Sackes, 2015, pp. 241-242)

In addition to content areas, science process skills were also included in the preschool standards (Sackes, Trundle, & Flevares, 2009). Eleven common science process skills that were most emphasized across the states were identified. These skills included asking questions (ten states; using basic tools (ten states); making observations (nine states); explaining cause and effect (nine states); making predictions (eight states); describing events and observations (seven states); collecting, organizing, and recording data (six states); communicating observations and findings (six states); ordering, sorting, and counting (six states); discussing and drawing conclusion (five states); and making comparison (five states) (Trundle & Sackes ,2015).

Accordingly, science education reform efforts should incorporate the expectations for preschool science education in the National Science Education Standards or the common core standards, and state-level reform efforts also should include standards for preschool science teaching and learning (Trundle & Sackes, 2012).

In spite of the significance of the reforms undertaken based on the Next Generation Science Standards (NGSS) (NGSS Lead States, 2013) or state level for early childhood science standards, there has been limited research in this area. Several interesting studies have been conducted on science education. For instance, Sackes (2014) examined parental priorities regarding science education in Turkey. The participants were 1,456 Turkish parents of pre-K children (36-72 months). The result demonstrated that the number of parents who prioritized science was quite low. The author suggested that parental preferences align well overall with early childhood teachers' tendency to teach less science than other content areas. Parental priorities might be major factor that contributes to the limitations in science learning experiences in students' early years.

Tippett and Milford (2017) incorporated STEM activities into the pre-k classroom. The methodology used observations, semi-structured interviews, focus groups, and a questionnaire to collect data from stakeholders. The purpose of this study was to examine how STEM activities were incorporated in pre-K classes, to explore students' engagement with STEM concepts, and to investigate parents' opinions about STEM. The findings supported the inclusion of STEM-based learning experiences for young children in Canada.

In Belgium, Dejonckheere and his colleagues tested the effects of integrating an inquirybased didactic method for teaching preschool science in an authentic practical classroom setting. Four preschool classrooms with children 4-6 years old participated in the experiment (N= 57). In order to assess children's attention to causal events and their understanding of concepts related to scientific reasoning skills, the study design involved a simple task in which a need for information was created. Compared to controls, children whose education had involved the so-called control of variables strategy showed significant learning gains in the posttest. Indeed, they carried out more informative and fewer uninformative explorations during their spontaneous play. The importance of such programs was discussed in relation to the field of STEM education (Dejonckheere et al., 2016).

The examples of research on science education in early childhood mentioned above demonstrate that early childhood science education research has been exploring the teaching of STEM subjects to young learners, parental involvement in the classroom setting, and the use of STEM activities in the classroom. Recently, early childhood science education for children in pre-k to 3rd grade classrooms, in particular, has been attractive to researchers who are engaged in investigating science teaching and learning.

Effective Early Childhood Science Instruction

The area of research on teaching science for young children is concerned with questions of what should be taught, how it should be taught, and for what ends or reasons. Young children are capable of developing considerable content knowledge about science, although this varies dramatically by child (Greenfield et al., 2009). Thus, young children can understand science concepts when they are presented with them in developmentally appropriate ways. Further, it is essential to provide accurate science content to young children to expand on their current knowledge of the world and to correct their misunderstandings. The National Association for the Education of Young Children (NAEYC, 2009) indicated that flexible instructional strategies were important for promoting young children's development from the infant/toddler stage to the early grades. According to the literature, the instructional strategies are the following:

- Fostering oral language and communication
- Drawing from a continuum of teaching strategies
- Making the most of the environment, schedule, and routines
- Setting up all aspects of the indoor and outdoor environment
- Focusing on children's individual characteristics, needs, and interests
- Linking children's language and culture to the early childhood program
- Teaching through social interactions
- Providing support for play
- Addressing children's challenging behaviors
- Supporting learning through technology

• Using integrative approaches to curriculum

Besides these instructional strategies advocated by the NAEYC, the literature also suggests that for early childhood science education, inquiry-based instruction is an effective early childhood science teaching approach (Dejonckheere et al., 2016; McLean, Jones, & Schaper, 2015; NRC, 2012; Trundle, 2009; Trundle & Sackes, 2015; Van Uum, Verhoeff, & Peeters, 2016). Inquiry-based instructional approaches offer the most effective way for young children to engage with and learn science concepts. Children are expected to be active agents in the learning activities, such as work in small groups, meaningful science activities, and sensemaking. One of the approaches that successfully incorporates inquiry into the teaching of science concepts is an approach based upon idea of "learning by doing." The Project approach, as it is known, involves an investigation of a topic in which children are interested (Curtis, 2002; Katz, 2018; Katz & Chard, 2000). Learning activities focus on child-generated questions guided by teachers. Children use various tools in their investigations, collect and analyze data, and produce artifacts (e.g., drawings, paintings, collages, and dramatic plays) to represent their understandings, and lead children collaborating with their peers (Trundle & Sackes, 2015).

Despite the advantages of the project approach, it involves some challenges. For instance, it is difficult to simultaneously balance children's interests with the required curriculum. Implementing a project approach is a time-consuming task that requires more work from teachers. Designing assessments to measure children's understanding that may develop from a project is also a challenge for many teachers (Curtis, 2002; Drake & Burns, 2004; Marx, Blumenfeld, Krajcik, & Soloway, 1997). In summary, despite the drawbacks, research literature and policy documents advocate for the use of a project approach in the teaching of science in

the early years (e.g., AAAS, 2001; Inan, Trundle, & Kantor, 2010; Morris, 2004; Trundle & Sackes, 2015).

Prospective Early Childhood Teachers' Conceptions of Science Education

Teachers' conceptions have been studied for decades. The term conception is used interchangeably with terms such as belief, orientation, idea, attitude, intention, and view (Antoniadou & Skoumios, 2013). Teachers' conceptions within science education literature will be referred to as teachers' conceptions of teaching and learning in science. Mahasneh (2018), defined teachers' conceptions of teaching and learning as "the beliefs held by teachers about their preferred ways of teaching and learning" (p. 819) including what teaching and learning mean, and the relationship between the teacher and students. Mahasneh (2018) mentioned that the conceptions of teaching and learning may be classified as traditional and constructivist. First of all, the traditional conception is based on teacher-centered methods where the teacher is the knowledge source and the student merely the passive recipient. By contrast, the constructivist conception sees knowledge as constructed by the learner, and the positive impact of a child's interaction with peers or adults is stressed (pp 531-532.).

Antoniadou and Skoumios (2013) described three dominant teaching approaches to educating students in science.

- The transmission approach sees knowledge as transferred from the teacher to the students, the main teaching practice is one in which students affirm scientific principles by carrying out experiments proposed by the teacher, and students work individually.
- In the discovery teaching approach, knowledge is not transferred from the teacher to the student, but rather knowledge is discovered by the student with the proper guidance of the teacher. Teaching strategies used in the discovery approach are experimentation, questions, investigation, and discussion (Fleer, 2009).

3. In the constructivist approach, knowledge is constructed by the learner. Students work collaboratively, as the process of constructing new knowledge passes between the social and the personal level.

Prior research on prospective elementary teachers' conceptions of science education has been carried out extensively; however, research on prospective early childhood teachers' conceptions of science education is limited. The research on prospective teachers' conceptions of science education has focused on primary teachers' conceptions about science teaching and learning (Ambusaidi & Al-Balushi, 2012; Antoniadou & Skoumios, 2013; Go & Kang, 2015; Subramaniam, 2013; Subramaniam et al., 2018).

Antoniadou and Skoumios (2013) investigated primary teachers' conceptions about issues related to science teaching and learning (the meaning of learning and teaching, the purpose of teaching, teaching tools, main teaching practices, classroom organization, and teachers' strategies to deal with students' errors in science). Teachers' conceptions are related to the views of the constructivist and discovery approaches. The findings of Antoniadou and Skoumios (2013) are similar to those of Go and Kang (2015), who investigated the self-images of science teaching held by early childhood prospective teachers who had taken constructivism early childhood science education courses. Draw a Science Teacher Test Checklist (DASTT-C), questionnaires, individual interviews, researchers' field notes, and participants' documents were collected. The results imply that early childhood science education should provide opportunities for pre-service teachers to reconstruct their own views about science teaching in order to learn and teach based on a constructivist view.

Brown and Englehardt (2017) provide insight into how a sample of early childhood preservice teachers who used iPads and apps in their coursework and high-stakes early learning

field placements made sense of using these devices as teachers. Their findings indicate instructional opportunities for teacher educators to consider as they seek to assist their students in making sense about how to implement as well as adopt appropriate and effective instructional strategies into their own classrooms. In-service or preservice work requires teacher educators to recognize that their students, more than likely, have limited experience with such devices as learners and/or as teachers.

Eckhoff (2017) documents a collaborative project involving preservice early childhood education students' development of inquiry-based learning experiences alongside kindergarten students within a science methods course. This study investigated the impact of an inquirybased teaching and learning experience on preservice teachers' beliefs about science in the early childhood classroom, utilizing a multiple method approach and data including classroom observations, transcripts from lesson planning sessions, PSTs' reflection journals, kindergarten science journals, and pre-and post-measurements for PSTs on the Draw-A-Scientist Test (DAST) and the Science Teaching Efficacy Beliefs Inventory—B (STEBI-B). Analysis of the DAST, the STEBI-B, and qualitative data revealed that the PSTs developed stronger understandings of inquiry-based science and self-efficacy beliefs related to their practice following participation in the collaborative project. Likewise, Lee & Yoon (2008) studied the teaching of prospective early childhood teachers, and how they assess children's inquiry process skills in a science method course. They found that such a course was essential for teachers to be able to help children understand the modes of inquiry and to foster their inquiry-based skills.

In summary, the literature on teachers' conceptions of teaching and learning science indicate the following.

- 1. Conceptions should be studied using appropriate qualitative research; in particular, the literate suggests using drawings and narrative.
- Conception can be studied by observing how teachers use iPads to assist their students in making sense about how to implement as well as adopt appropriate and effective instructional strategies into their own classrooms.
- Conceptions can be studied in relation to prospective early childhood teachers' development of inquiry-based learning experiences and inquiry skills within a science methods course.

The Role of Drawing in Educational Research

Student drawings in science class can be used as a way to document educational phenomena in relation to science education. Activities such as drawing images and doing narrative writing about teaching science can be a means for exploring teachers' beliefs about science education (Thomas et al., 2001). Drawing enables many people to express their inner thoughts, which they often cannot do through written or narrative texts (Markic & Eilks, 2008). Clearly, drawings can contribute to providing evidence in science education as they enable researchers to gain insights into teachers' thinking about science teaching and learning; to gain insights into learners' thinking about science teaching and learning; to study personally-generated drawing as a methodology; to gain insights into teachers' thinking about science education policy; to gain insights into teachers' thinking about science content; and to gain insights into learners' thinking about science content; and to gain insights into learners' thinking about science content; and to gain insights into learners' thinking about science content; and to gain insights into learners' thinking about science content; and to gain insights into learners' thinking about science content; and to gain insights into learners' thinking about science content (Project Nexus, 2011).

The literature on the use of drawing looks at its use in terms of developing both a model for inquiry and a method of assessment for prospective science teachers, for studying prospective teachers of elementary education, for developing a methods course for graduate students specializing in elementary science education, and for designing a science content
course for early childhood teachers. Existing research has been conducted on using drawing for undergraduate students in teacher education programs of elementary education, secondary education, or higher education, and for prospective science teachers (Ambusaidi & Al-Balushi, 2012; Katz et al., 2011; Markic & Eilks, 2013; Minogue, 2010; Project Nexus, 2011; Sinclair et al., 2013), but little research has been conducted examining the effects of teacher preparation programs on early childhood teachers (Mckinnon & Perara, 2015).

In a study of elementary and secondary education students enrolled in a required introductory education course, Sinclair and her colleges examined participants' visual images of themselves as teachers, and they modified the Draw-a-Science-Teacher Test (DASTT-C) (Thomas et al., 2001), to create the Draw-a-Teacher Checklist instrument (Sinclair et al., 2013). The results indicated that the graduate participants viewed teachers and teaching in a more teacher-centered way. This suggests that these teachers tend to revert back to more traditional, teacher-centered classroom beliefs, and they often indicate frustration in the classroom. The study further investigated the reasons why some teachers return to teachercentered classroom practices even when they are aware of the effectiveness of studentcentered classroom practices and the need to use a balanced approach in the classroom.

Ambusaidi & Al-Balushi (2012) explored the beliefs of prospective science teachers in the College of Education at Sultan Qaboos University/Sultanate of Oman about science teaching by using the Draw-A-Science-Teacher-Test Checklist (DASTT-C) tool in a Science Methods course. The results revealed that after completing the Science Methods course, prospective science teachers shifted significantly from a teacher-centered approach to an intermediate state between the teacher-centered and student-centered approaches.

Additionally, Minogue (2010) documents the use of the DASTT tool to analyze prospective elementary teachers' beliefs about science teaching and science methods courses. Results indicated statistically significant shifts in participants' mental models of science teaching and learning. The post-course stage showed that more students adopted student-centered practices. The use of the Draw-a-Science-Teacher-Test as a diagnostic tool for both preservice teacher beliefs about science teaching and science methods courses is effective.

Markic and Eilkic (2013) conducted an international cross-level study of German prospective teachers' beliefs about teaching and learning chemistry. The study was based upon drawings of teaching situations, which were analyzed using an evaluation pattern developed using grounded theory. Qualitative scales were used to analyze beliefs about classroom organization, teaching objectives, and epistemological beliefs. Data were collected from university freshmen, student teachers midway through their university teacher education program, and recently graduated teachers who had just finished their university program and were gaining experience as full-time teachers. The initial results revealed that the freshmen in their study professed very traditional beliefs about teaching and learning (characterized by teacher-centeredness and an understanding of learning as receptive consumption).

The Science Teaching and Learning Portrayals of Professional Practices (STLP3)

This current study used the instrument called the Science Teaching and Learning Portrayals of Professional Practices (STLP3). With this instrument, drawings are analyzed using an evaluation pattern based on four strands of the Science Teaching and Learning Portrayals of Professional Practices (STLP3) Scoring Rubric (Katz et al., 2011). Project Nexus blended formal and informal science education (Project Nexus, 2011), and this instrument was adapted by

Project Nexus to be used with prospective teachers to investigate their teaching of science in an

early childhood classroom (Mckinnon & Perara, 2015).

The initial STLP3 instrument is based on the six statements of the NRC (2009) document,

that "Learners who engage with science in informal environments....

- 1. Experience excitement, interest and motivation to learn about phenomena in the natural and physical world.
- 2. Come to generate, understand, remember and use concepts, explanations, arguments, models, and facts related to science.
- 3. Manipulate, test, explore, predict, question, observe, and make sense of the natural and physical world.
- 4. Reflect on science as a way of knowing; on processes, concepts, and institutions of science; and on their own process of learning about phenomena.
- 5. Participate in scientific activities and learning practices with others, using scientific language and tools.
- 6. Think about themselves as science learners and develop an identity as someone who knows about, uses and sometimes contributes to science." (NRC, 2009, p. 43)

Project Nexus was based on qualitative methodology. Thus, the teachers' drawings provided useful data for examining statements 1, 2, 3, and 5. Statements 4 and 6 are more directly amenable to study through written and verbal data, and they are not easily illustrated or analyzed using drawings (Project Nexus, 2011; Mckinnon & Perara, 2015). For the current study, the Science Teaching and Learning Portrayals of Professional Practices (STLP3) Scoring Rubric (See Appendix B) was used to examine the specific goal that was presented in the drawing. The drawings were subsequently scored from 0 to 16 based on four strands as follow:

- 1. *Affective*: Students should experience excitement, interest, and motivation to learn about phenomena in the natural and physical world.
- 2. *Cognitive*: Students will come to generate, understand, remember, and use concepts, explanations, arguments, models, and facts related to science.

- 3. *Implementation*: Students must manipulate, test, explore, predict, question, observe, and make sense of the natural and physical world.
- 4. *Social*: Students should participate in scientific activities and learning practices with others, using scientific language.

According to Project Nexus (2011), a methodology using drawings could be beneficial to workers in the education research community. The project reported that the use of drawings as a data collection method was useful for gaining insights into prospective teachers' thinking that had not been clear previously in written or verbal data; engaging in discussion in interviews and emails about insights gathered from the drawings of the prospective teachers; encouraging prospective teachers to reflect on aspects of their developing teacher identities that they had not previously perceived; and giving the teacher candidates a mode of expression that they enjoyed and was novel to them as a data collection strategy (Project Nexus, 2011).

The Sociocultural Perspective

In the present study, the students' drawings and narratives will be analyzed based on four strands of the Science Teaching and Learning Portrayals of Professional Practices (STLP3) Scoring Rubric (Katz et al. 2011). One strand of this instrument examines the "social" aspect by analyzing the drawings in terms of participation in scientific activities and learning practices with others, using the scientific language and tools. The prospective teachers' narrative writing will be used to identify the theme of teaching strategies to reveal students' interaction with others in scientific activities and learning practices. Consequently, sociocultural theory is appropriate in the current study because it aligns with the use of the STLP3 for analysis of drawings. Further, sociocultural theory is appropriate in relation to the use of teachers' narratives to study scientific activities in which students participate to learn along with others and to measure prospective early childhood teachers' conceptions of science instruction in the early childhood classroom.

The sociocultural approach of Vygotsky emphasizes the inextricable link between the individual and the social, where the activities of the individual are always culturally contextualized. Sociocultural theory literature falls into two types. One is that promoted by the Russian psychologist Ley Vygotsky (1896-1934). Another type is the literature produced by interpreters of Vygotsky's sociocultural theory (Neo-Vygotskians). Different from Piagetian constructivism and radical constructivism, sociocultural theory pays serious attention to three crucial aspects: the structural characteristics of schooling; the social organization of instruction; and cultural tools (material and psychological), especially the tools of language and inquiry (John-Steiner & Mahn, 1996; Subramaniam, 2007).

Sociocultural perspective emphasizes the teacher's role within students' construction of meaning. The teacher's role is cited as a mediator of students' learning. The key role of teachers is to motivate students to be active members and contributors within social contexts mediated by cultural tools (material and psychological tools) (Blanck, 1990; Miller,1996; Subramaniam, 2007). Teachers have an important role to play in a child's developing ability to think and to think well. The dialectic connection between cultural tools, interpersonal relations, and students' practical activities transforms students' natural cognitive functions into cultural ones. As can be seen, within this theory, teaching is assisted performance, where assistance is provided at points within the students' zone of proximal development.

Accordingly, knowledge is not generated from within; rather, learning stimulates and leads to development. Culture and social interaction collaboratively guide cognitive

development. The zone of proximal development is created by the teacher and student in the context of specific tasks for intersubjective agreement about meaning being made together: it is not simply announced by the teacher but is carried out through mutual participation. This theory captures the idea of teachers as experts in the use of cultural tools leading to cultural amplification and the extending of students' cognitive processes. This study incorporates psychological tools (the semiotic tools of speech, writing, language, and thought) and physical tools (students' drawings and narratives) which mediate thought and activity within the zone of proximal development.

Another key construct of this theory that is particularly relevant to the present study is the relationship between language and thought within the zone of proximal development. In sociocultural theory, thought is expressed through language. To illustrate, the teachers act through metaphors and other figures of speech. They use these as tools to shape and specify their relations to students and their teaching circumstances: a repository of linguistic artifacts for the use in meaning making (Hodson & Hodson, 1998a; 1998b; Howe, 1996; Kozulin, 1990; Miller, 1996; Subramaniam, 2007). In the present study, teachers gave the directions to the students, then the students expressed their thoughts in response to drawings and narratives.

Another essential element of social interaction is scaffolding. In Vygotsky's view of cognitive development, the adults or other partners in a child's world provide scaffolding to help children learn new information and develop more complex thinking abilities. In the classroom, the activity structure manifests as the "interactive opportunities" that enable the teacher to select the relevant and suggested content, organize the content, and relate it to what their audience already knows. As a result, the teachers enable themselves and their

students to "publicly display thinking and reasoning for their students to gain access to learning and to demonstrate social and academic competence" (Subramaniam, 2007; Weade, 1987: 17).

Subramaniam (2007) clarified that the zone of proximal development reflects the historical, social, and cultural context in which the cultural templates of society are made intersubjective for students to create meaning. His work also indicates that language, thoughts, and actions within the constructs of paradigmatic and narrative modes of thought reflect multiple ways of viewing reality within this zone of proximal development: that is, they are symbolic devices. Thus, language, thought, and actions, expressed within narratives, are key analytical templates to view the meaning-making processes within the zone of proximal development.

On the other hand, limitations of using the linguistic and psychological aspects of student and teacher interactions within the zone of proximal development are cited by some researchers. Namely, an over-emphasis on linguistic competence may lack conceptual depth (Frawley, 1997; Miller, 1996; Subramaniam, 2007). However, linguistic and psychological aspects must be taken into account with ample consideration of the social context of interaction in which they occur, as shown by Neo-Vygotskians (Confrey, 1995; Tudge & Winterhoff, 1993; Subramaniam, 2007). On the one hand, Subramaniam (2007) recommends that the inhibitory functions of language and its boundaries be captured within the analysis of language in social contexts.

A summary of these major constructs pertaining to teachers and teaching as perceived within sociocultural theory is presented in Figure 2. The figure is adapted from Carter et al. (1999) and captures the key constructs of sociocultural theory and connects these constructs

into a composite figure that reflects teaching and learning within the zone of proximal development. It serves as the framework for the current study.



Figure 2. A diagrammatic representation of researcher's articulation of the sociocultural theory

The Sociocultural Perspective on Science Education

Sociocultural learning theory has been advocated and promoted in teaching and learning in many disciplines, and it suggests that science should not be separated from the society and culture in which teaching and learning in science education are situated (Erdogen, 2016). The sociocultural perspective on science education did not take root until after 1970, and it became associated with the "discovery" of Vygotsky's ideas on constructivist learning (Hassard & Dias, 2009). Children construct knowledge through social interactions (Santrock, 2004). Vygotsky suggests that emergent literacy develops through these social interactions in which children develop knowledge, values, and skills. These interactions differ from child to child, which leads to diverse development (Nord, Lennon, Liu, & Chandler, 2010).

Vygotsky questioned how children learned science from the historical and sociological contexts in daily life. He believed that children's informal daily interactions provided a bank of experiences to draw on to develop more formal, scientific, and conceptual knowledge in later schooling (Hedges et al., 2011). Accordingly, the focus of the sociocultural perspectives on

science education includes the role of social interaction in teaching and learning science or in studying the world, whether in classrooms or research laboratories; interpersonal social interaction, whether collaboration in a laboratory or dialogue in a classroom; and sociocultural theory (Lemke, 2001).

The social context for learning science with peers in an early childhood classroom is crucial. Science is a process, and it almost always involves collaboration with others. The Next Generation Science Standards (NGSS) are built on the key assumption that "science is fundamentally a social enterprise, and scientific knowledge advances through collaboration and in the context of a social system with well-developed norms" (NGSS Lead States, 2013, p. 27). Clearly, learning with peers has been identified as a critical component of science learning and science education, with science standards highlighting the importance of providing students opportunities for collaboration, discussion, and reflection around science.

Social interaction and peer collaboration during science learning opportunities may contribute to children's learning of science and inquiry. For instance, children may benefit from jointly planning of an investigation, raising questions through discussion with peers, describing outcomes and patterns in data, and explaining, sharing, discussing and listening as they work with peers on scientific investigations. It has also been argued that working with peers on science problems exposes children to a variety of forms of thinking and interpersonal interactions that likely benefit children both socially and academically (Johnson-Pynn & Nisbet, 2002).

What cultural practices relate to science skills learned in science education in early childhood settings? How can teachers bring these cultural practices of children to science

classrooms? According to Bransford and his colleagues, science instruction in early childhood emphasizes learner-centered environments characterized by the ability of the teacher to pay careful attention to the knowledge, skills, attitudes, and beliefs that learners bring to the classroom and use these experiences to structure and organize the learning experiences of children (Bransford, Brown, & Cocking, 2000). Children build knowledge by using both general domain and specific domain learning processes and benefitting from support from older children and adults (Hong, Torquati & Molfese, 2013).

Teachers play the role of understanding what the child is thinking and the child's misconceptions and readjusting the learning environment to enable the child to figure out the solution to the problem at hand, as they pursue a science activity. Culture is integrated in teaching approaches with a learner-centered environment. Teachers will integrate personal and social perspectives on learning and emphasize the role of dialogue in helping students construct science knowledge. Integrating culture into the instruction is challenging for educators in recent years, especially for Early Childhood Science Education.

Therefore, teaching approaches referred to as "culturally responsive," "culturally appropriate," "culturally compatible," and "culturally relevant" are important in classrooms (Bransford, Brown, & Cocking, 2000). Culturally responsive teaching is a method that engages teachers in identifying teaching and learning approaches related to integrating culture into the curriculum (Sleeter, 2010). Strategies of teaching science that emphasize observation, listening to children, questioning, helping children make connections, and learning by making mistakes support the concept of diagnostic teaching or learner-centered teaching.

Materials for teaching science should originate from the natural environment of the children and the families. Cans, plastic cups or bottles can be used in science lessons about the recycling or used for creating telephones. The important calendar days of each country can also be integrated in science lessons for children. For example, teachers can teach about the weather on Song Kran days and ask about the culture in Thailand. Children will gain science knowledge and skills relevant to their family's culture. Relating to a research study of Ng'asike (2010), targeting the teaching of science to match the cultural experiences of children ensures that knowledge and skills of science are meaningful and relevant to the life of the learners.

The sociocultural perspective is part of Development Appropriate Practice (DAP), an important concept relating to the early childhood science curriculum. Kostelnik, Soderman, and Whiren (2011) conclude that teachers must look at children and families within the context of their community and culture before creating meaningful learning for children. Providing young children with opportunities to see science in their everyday activities helps them to build the basic understandings and interest for future science learning (Lind, 1998). Lack of cultural awareness by early childhood professionals can lead to erroneous assumptions and potentially negative outcomes for children.

Although the sociocultural framework has been quite extensively researched and utilized in comprehending students' learning in a variety of disciplines, the idea of prospective early childhood teachers' conceptions in teaching science in a socially constructed way has been minimally investigated. As one example, Eckhoft (2017) investigated the impact of an inquiry-based teaching and learning experience on prospective early childhood teachers' beliefs about science in the early childhood classroom. This study investigated the ways in which

prospective early childhood teachers incorporated visual arts media into inquiry experiences to encourage young children to visually represent their scientific thinking. Socio-constructivist philosophy was used as a theoretical framework in addition to the long tradition of inquirybased science instruction. The findings indicate that prospective teachers of early grades need direct experience with inquiry-based teaching in order to appreciate the strengths and challenges associated with the approach. Prospective early childhood teachers gain experience as both learners and teachers. They serve as a source of support as they seek to appropriate inquiry-driven pedagogical practices within the pedagogically restrictive environments of many contemporary early childhood classrooms.

Providing prospective early childhood teachers with challenging and supportive inquirybased teaching and learning opportunities during their teacher preparation program can ultimately assist in strengthening their understanding of their role as supportive educators in developing science learning experiences. Based upon current research recommendations for best practices in the classroom, therefore, it is crucial to conduct research based on sociocultural theory in science teaching for prospective early childhood teachers.

Summary

This chapter examined the literature concerning prospective early childhood teachers' conceptions of science instruction, presented the standards that inform the content and practice of early childhood science education, discussed the current instructional strategies, and explained the role of drawings and other instruments that will be used to analyze the data collected for this study. Lastly, the sociocultural perspective that will be used as the theoretical framework to guide and frame the study was discussed. Chapter 3 identifies the methodology,

the research design, the population studied, the data sources and data collection, and the data analysis employed for the present study. Chapter 4 presents the findings of prospective early childhood teachers' conceptions of science instructions from drawings and narratives. Next, Chapter 5 discusses science instruction for an early childhood classroom as it relates to the findings presented in Chapter 4. It then offers implications for future study.

CHAPTER 3

METHODOLOGY

The purpose of this study was to investigate prospective early childhood teachers' conceptions of science instruction in an early childhood education methods class. The primary research question for this study was this: What are prospective early childhood teachers' conceptions of science instruction in an early childhood classroom? This Chapter presents the research design, gives details on the selection of participants, and describes the rationale and processes by which the data were identified, gathered, and managed. Information about the instruments and procedures used for data analysis is included.

Research Design

The current study conducted an analysis of prospective early childhood teachers' drawings and narratives to identify their conceptions of teaching science. The early childhood teachers' drawings and narratives were collected using a qualitative research design. Qualitative research is an approach that encompasses multiple theoretical paradigms, methods, and approaches, such as case study, participatory inquiry, interviewing, participant observation, visual methods, and interpretive analysis (Hesse-Biber, 2017). The qualitative researcher is an "interpreter" using a "naturalistic approach" to construct meaning from individuals' complex lived experiences and realities. The challenge for a qualitative researcher is constructing the research study, in particular choosing the appropriate tools/methods. The methodology for the current study was adapted from Hesse-Biber (2017), as shown as Figure 3.



Figure 3. The purpose of methodology.

Participants

Participants for this study were drawn from 415 prospective early childhood teachers who self-enrolled in two of the required early childhood education courses as part of a teacher preparation program. Data were gathered in the fall of 2016, the spring of 2017, the summer of 2017, and the fall of 2017 at a large university in the southwest region of the United States. All participants were sophomores and juniors working toward certification in early childhood through sixth grade with bilingual supplemental, ESL supplemental, and special education EC-12 programming. As shown in Table 1 below, the participants consisted of 392 females (94.5%) and 23 males (5.5%). The racial/ethnic composition was as follows: 220 Caucasians (56%), 44 African Americans (11%), 105 Hispanics (27%), 21 Asian Americans (5 %), and two multi-racial participants (1%). A sample of 100 prospective teachers was selected from among the 392 prospective teachers (See Table 1). Figure 4 presents a diagram showing the number of participants by race and ethnicity.

Table 1

Number of Participants by Race and Ethnicity

Race/Ethnicity	Total Sample	Total Females	Selected Study of Participants
White	234	220	56
Hispanic	112	105	27
Black/African American	46	44	11
Asian	21	21	5
Native American/Pacific Islander	2	2	1
Total	415	392	100





The selection of 100 participants from the larger sample of 415 participants was based on: (a) participants' self-identification of ethnicity, and/or race, (b) participants' selfidentification of their gender as female, (c) each participant having a drawing that was accompanied by a five to ten sentence narrative, and (d) clarity of participants' drawings.

Research Context

The current study was contextualized in a teacher preparation program. All participants were enrolled in two early childhood education courses. The teacher preparation program in which they were enrolled was accredited by the Council for the Accreditation of Educator Preparation (CAEP) and the Texas Education Agency (TEA). The participants' course background for the teacher preparation program in early childhood through grade six (EC–6) included core subjects with the following certification options: teacher certification with no specialization certificate, ESL supplemental certification, bilingual supplemental certification, and special education EC–12 certification, as shown in Figure 5 below. Participants' courses in three core areas: the Science and Mathematics core, the Teacher Education core, and the Early Childhood core (See Figure 5).

Data Collection and Data Sources

The current study carried out an analysis of data collected for a larger study. Two items of data, a drawing and a narrative, were collected from each participant on the first day of class in the two courses selected, in the fall of 2016, the spring of 2017, the summer of 2017, and the fall of 2017. The prompt "Draw a teacher teaching science in an early childhood classroom" was given to each participant. The goal of the larger study was to gain insights into prospective early childhood teachers' conceptions of science teaching. In addition to responding to the prompt, each participant was instructed to write a narrative describing her own drawing in five to ten sentences.



Figure 5. Example of participants' course background in a teacher preparation program.

Drawing Procedures

Participants were provided with a blank sheet of white paper and drawing supplies (markers, color pencils, and crayons) and were asked to respond to the prompt "Draw a picture of a teacher teaching science in an early childhood classroom." No other instructions were provided, and participants were given approximately 15-20 minutes to complete the drawing. The purpose of using drawing in this study was to collect participants' images and investigate their conceptions of teaching science in an early childhood classroom. Accordingly, the drawing served as a potential instrument—representing projective research techniques, pictorial forms, symbols, forms of text, markers and mirrors, and platforms to elicit discussion—that visually captured participants' interactions with a phenomenon not easily illustrated through text or narratives (Subramaniam et al., 2018). In the field of educational research, drawings have been used to illustrate participants' experiences, tacit knowledge, and practical knowledge about instruction (Brown & Schwartz, 2009; Subramaniam et al., 2018). Therefore, the drawings revealed aspects of the science teaching identities of teachers of early childhood learners.

Narratives

In the present study, narratives were also collected as a data source. Narratives were collected with two objectives: (a) to provide an image and text balanced approach (Radnofsky, 1996; Subramaniam et al., 2018), and (b) to allow participants to describe and interpret their own drawings in their own words (Subramaniam et al., 2018). As qualitative data, narratives can be seen as data reflected in participants' accounts of phenomena, collected in the participants' natural speech (Connelly & Clandinin, 1986; Subramaniam et al., 2018). In this

study, the prompt used to collect narrative data was "Briefly describe your drawing in five to ten sentences. Specifically, describe the products and the processes in each drawing."

Data Analysis

Analysis of Drawings

Participants' drawings of their conceptions of teaching science were analyzed based on four goals of the *Science Teaching and Learning Portrayals of Professional Practices* (STLP3) Scoring Rubric (Katz et al., 2011). The original instrument was based on the goals (or strands) for science learning from the Taking Science to School (NRC, 2007) report. The later Learning Science in Informal Environments (NRC, 2009) report increased the number of goals to six, and it was on these goals that the STLP3 was based (Katz et al., 2011). Goals four and six of the original list were omitted as their analysis relies on verbal and written responses (Project Nexus, 2011).

The goals state that "Learners who engage with science in informal environments. . .

- 1. Experience excitement, interest and motivation to learn about phenomena in the natural and physical world.
- 2. Come to generate, understand, remember and use concepts, explanations, arguments, models and facts related to science
- 3. Manipulate, test, explore, predict, question, observe and make sense of the natural and physical world.
- 4. Reflect on science as a way of knowing; on processes, concepts, and institutions of science; and on their own process of learning about phenomena.
- 5. Participate in scientific activities and learning practices with others, using scientific language and tools.
- 6. Think about themselves as science learners and develop an identity as someone who knows about, uses and sometimes contributes to science. (NRC, 2009, p. 43)

Participants' drawings provided useful data for goals 1, 2, 3, and 5, and the STLP3 (See Appendix C) was adapted to analyze the prospective teachers' teaching of science at early childhood/primary school levels in a formal setting (Mckinnon & Perara, 2015). As the current study included only drawings and written responses, analysis of goals 4 and 6 was not feasible, and thus goals 4 and 6 were excluded. To illustrate the analytic procedure, each goal of the Science Teaching and Learning Portrayals of Professional Practices (STLP3) Scoring Rubric examined a specific goal that was presented in the drawing. The drawings were subsequently scored from 0 to 16 based on the four goals, and each goal was scaled from 0 to 4 (See Appendix B). The goals of the STLP3 are as follows:

- 1. *Affective:* Students will experience excitement, interest and motivation in learning about phenomena in the natural and physical world (Goal 1).
- 2. *Cognitive*: Students will come to generate, understand, remember, and use concepts, explanations, arguments, models, and facts related to science (Goal 2).
- 3. *Implementation*: Students will manipulate, test, explore, predict, question, observe, and make sense of the natural and physical world (Goal 3).
- 4. *Social*: Students will participate in scientific activities and learning practices with others, using scientific language (Goal 5).

In addition to the STLP3, a Scoring Rubric Supplemental Information Sheet (See Appendix C) was used to clarify specific indicators of each goal. Examples include for *Affective* (goal 1), "...Look at the mouths of the figures present. If anyone is smiling, give credit (that is, if only the teacher is smiling but the students are not, give credit for smiling or the reverse) ..." *Cognitive* (goal 2), "Identify concepts, explanations, arguments, models, and facts using these descriptions—identify concepts and add thought bubbles or comments about bigger science ideas (e.g. energy, evolution)." *Implementation* (goal 3), "Identify manipulating, testing, exploring, predicting, questioning, observing, and sense-making using these descriptionsmanipulating - each learner has access to materials in reach or is shown actually touching items (note: manipulating variables for an experiment)" *Social* (goal 5), "Identify participation in scientific activities and learning practices with others, using scientific language, and tools using these descriptions; participate in scientific activities and learning practices with others students are grouped for interaction."

Analysis of Narratives

Participants' narratives were analyzed to reveal conceptions of science instruction in an early childhood classroom. The data analysis procedure used an inductive approach consisting of category development to analyze narratives. The process of data analysis began with transcribing the participants' narratives. First, the participants' narratives were labeled through the highlighting of words and/or single phrases in the data and then, through the analytical process of interpretation, the data were assigned codes by the researcher and one member of the dissertation committee. Coding is a means of organizing qualitative data by summarizing, reducing, or condensing it, in order to develop concepts, topic, ideas, categories, and themes (Coffey & Atkinson, 1996). Specifically, coding allows the researcher to identify meaningful data that can be used in the analytical process of inference and interpretation. Hahn (2008) cites three levels of qualitative coding:

- 1. Initial Coding or Open Coding, where large quantities of research data are focused and labeled;
- Focused Coding or Category Development, where reexamination of level one codes and data is further focused on;
- 3. Axial or Thematic Coding, which results in the development of themes.

Second, after coding the data, the authors compared these codes to identify themes and consistent patterns as they emerged through the data collection. These patterns helped to answer the research questions: What are prospective early childhood teachers' conceptions of science instruction in an early childhood classroom? This served as a starting point for manual coding. Manual coding consisted of several readings of participants' narratives. During this procedure, it was found that all coded text appeared to describe teaching strategies, and some categories were revised, removed, and added. Data that could not be coded into one of the predetermined categories were coded with new categories. Then the initial codes were examined and compared for patterns, and similar codes were organized into categories so that meaning could be derived from the data (Saldaña, 2016). The categories were reexamined for major themes (Teaching Strategies) present in the data. Saldaña (2016) states the following, "a theme is an outcome of coding, categorization, and analytic reflection, not something that is, in itself, coded. . ." (p. 13). In this present study, together with one member of the dissertation committee, the researcher reviewed the data and codes with the research study question in mind and identified six major emerging themes of teaching strategies from the analysis of participants' narratives: demonstration, engagement/enjoyment, inquiry/interaction, experiment, safety, and observation.

Next, the number of times each theme was identified in the data was quantified, and a percentage was used to represent the frequency of that theme. Lastly, the findings from the analysis of data (the participants' drawings and narratives) were peer-debriefed with an expert in early childhood teaching. The expert had six years of pre-K teaching experience and 35 years as an instructor in early childhood teacher education and administration. At the time this study

was conducted, the expert was the Director of the Child Development Laboratory at the study site. During the peer debriefing, the expert presented and shared her insights about the findings from the analysis of both the participants' drawings and the narratives. All correspondence with the expert was recorded and was used to re-examine and reanalyze the findings and the data. For example, scores from the STLP3 on cognitive and implementation dimensions were reexamined, and in the subsequent analysis the percentage for the cognitive dimension dropped from 65% to 33%, and for the implementation dimension, the scores dropped from 65% to 37%. Also, the number counts for participants expressing different conceptions of science instruction were reexamined with the narrative data, and in the subsequent analysis the percentage for each conception was recalculated. Figure 6 below presents the analysis of data framework of the present study.



Figure 6. Data analysis framework.

Following this description of the analytic process, the results will be described in sufficient detail so that readers will have a clear understanding of how the analysis was carried out and its strengths and limitations (GAO, 1996; Elo & Kyng's, 2007).

Trustworthiness

Trustworthiness in this study relies on the criteria developed by Lincoln and Guba. Trustworthiness aims at supporting the argument that the study's findings are "worth paying attention to" (Lincoln & Guba, 1985; Elo et al., 2014). The criteria for evaluating trustworthiness in the current study include credibility, transferability, dependability, confirmability, and authenticity measures (Elo et al., 2014). Table 2 presents the qualitative criteria employed for assessing research quality and rigor in the current study.

Table 2

Trustworthiness	Strategy Employed	
Credibility	Use of peer debriefing Co-analysis	
Transferability	Providing thick description	
Dependability	Code-recode strategy	
Authenticity	Researchers, fairly and faithfully	

The Qualitative Criteria for Assessing Research Quality and Rigor

Researchers must ensure that those participating in research are identified and described accurately; this addresses the *credibility* of the study. The term *dependability* refers to the stability of data over time and under different conditions. In this study, dependability was achieved by *confirmability*, which relates to the objectivity, that is, the potential for congruence between two or more independent people about the data's accuracy, relevance, or

meaning. The term *transferability* refers to the potential for extrapolation. It relies on reasoning supporting the idea that findings can be generalized or transferred to other settings or groups. Finally, the term *authenticity* refers to the extent to which researchers, fairly and faithfully, show a range of realities (Lincoln & Guba, 1985; Polit & Beck, 2012; Elo et al., 2014).



Figure 7. Steps to ensure trustworthiness.

In the present study, several steps were taken to ensure the trustworthiness of the findings. As presented in Figure 7 above, adapted from Subramaniam et al. (2018), all data, including the drawings and narratives, were coded individually by the researcher and one of the

dissertation committee members, and an intercoder agreement process (Kurasaki, 2000) was carried out to seek consensus. The drawing data were analyzed by the researcher and by the participants themselves, who wrote the accompanying narratives of the drawings. The use of only drawings has limitations, such as the inability of drawings to capture every detail and nuance of the phenomenon: the phenomena of artistic aptitude/artistic plateau may be relevant. Therefore, the researcher sought to use narratives and intercoder agreement to achieve consensus on the findings.

In addition, demonstrating transferability, findings from this study can be used in future studies comparing prospective early childhood teachers' conceptions of science instruction for young learners in the United States with those of prospective early childhood teachers in other countries.

Summary

The methodology of this study was designed for the investigation of prospective early childhood teachers' conceptions of science instruction. The study utilized participants' drawings and their narratives of their drawings. Four strands of the Science Teaching and Learning Portrayals of Professional Practices (STLP3) Scoring Rubric were used to analyze participants' drawings, and qualitative analysis was applied to analyze narratives. This process served to identify participants' conceptions of their science teaching identities for teaching early childhood learners.

CHAPTER 4

FINDINGS

The purpose of this study was to investigate the prospective early childhood teachers' conceptions of science instruction in an early childhood classroom context. This chapter reports the findings from analysis of participants' drawings and participants' narratives of drawings of science instruction in an early childhood classroom. The findings are presented in two sections: the first section presents participants' conceptions of science instruction based on the analysis of participants' narratives of drawings.

Participants' Conceptions of Science Instruction: Drawings

Analysis of participants' drawings using the STLP3 instrument revealed four themes that characterized participants' conceptions of science instruction. These four conceptions were: (a) science instruction in an early childhood classroom is a social activity, (b) science instruction in an early childhood classroom as an affective endeavor, (c) science instruction in an early childhood classroom is the implementation of activities for learning, and (d) science instruction in an early childhood classroom is a cognitive activity. Table 3 presents the findings from the analysis of the 100 participants' drawings conceptualizing science instruction for children in pre-K to 3rd grade. This table details the number of times these four conceptions appeared in the drawing data and it also lists the percentage of the total represented by each conception.

As is evident in Table 3, 85% of the participants' conceptions of science instruction were within the social dimension. That is, participants conceptualized science instruction in a pre-K to 3rd grade early childhood classroom as being a social activity. Participants' conceptions of

science instruction was coherent with the STLP3's description of science instruction as students participating "in scientific activities and learning practices with others, using scientific language and tools." Additionally, 73% of participants conceptualized science instruction in a pre-K to 3rd grade early childhood classroom as an affective endeavor. That is, this conception was coherent with the STLP3's description of science instruction as being underscored by "experience excitement, interests, and motivation" (Project Nexus, 2011) of both teacher and learners, revealing the way they learn about phenomena in the natural and physical world.

Table 3

Scores and Percentages of Participants' Conceptions of Science Instruction by Dimensions Present in the STLP3 Rubric (N = 100)

Conceptions	Total of Score of each Goal (400)	Percentage(%)
Affective	293	73
Cognitive	132	33
Implementation	146	37
Social	340	85

On the other hand, 37% of participants had conceptions of science instruction as implementing activities for learning. Participants' conceptions of science instruction were coherent with the STLP3's description of science instruction as implementing activities for learning, that is, students "identify manipulating, testing, exploring, predicting, questioning, observing, and sense-making of the natural and physical world." Likewise, 33% of participants had conceptions of science instruction as a cognitive activity. Participants' conceptions of science instruction were coherent with the STLP3's description of science instruction as cognitive activity, that is, students "come to generate, understand, remember and use concepts, explanations, arguments, models and facts related to science."

Next, I present examples of each conception using participants' drawings. For example, Figure 8 details a participant drawing that received a score of four showing evidence of learning with others, using scientific language and using scientific tools. Figure 9 shows an example of a participant's drawing that portrays no social activity. Figure 10 illustrates a participant drawing that received a score of four showing "smiling figures with specific indicators." Figure 11 shows an example of a participant's drawing that portrays no affective endeavor. Figure 12 presents a participant drawing with a score of four showing science instruction as an implementation of activities for learning. Figure 13 shows an example of a participant's drawing that shows no evidence of an implementation of activities for learning. Figure 14 details a participant drawing that received a score of four showing "evidences or models of concepts, explanations, arguments, models, or facts." Figure 15 shows an example of a participant's drawing that shows no evidences of cognitive activities for learning.



Figure 8. A participant's drawing receiving a score of 4 for science instruction as a social activity.



Figure 9. A participant's drawing receiving a score of 0 for science instruction as a social activity.



Figure 10. A participant's drawing receiving a score of 4 for science instruction as an affective endeavor.



The teacher in my drawing is holding a test tube with, some vineger in it. In her lesson she is teaching she wants to teach her students the difference between a beak lest tube, gradwated cylinder, ed. There are it various liquids inside the tubes she wants the students to please turn over the page it additional space is needed communicate the differences/compace/contrast all of the liquids. She tells the students what is in each glass tube. The students have to work in groups.

Figure 11. A participant's drawing receiving a score of 0 for science instruction as an affective endeavor.

:



Figure 12. A participant's drawing receiving a score of 4 for science instruction as an implementation of activities for learning.



Figure 13. A participant's drawing receiving a score of 0 for science instruction as an implementation of activities for learning.


Figure 14. A participant's drawing receiving a score of 4 for science instruction as a cognitive activity.

aren a Haunfr inat believes ALD AN APH 1 ANIN WELLY - (1 to ed snoes o N () she is nappy to see har shu

Figure 15. A participant's drawing has a score of 0 for science instruction as a cognitive activity.

Figure 16 shows that the teacher and students are smiling. It includes specific indicators such as the use of the descriptive words "fun" (which is affective). The drawing shows evidence in thought bubbles, containing words such as vocab, hand on, model of concept (physical science, a literacy component) (a cognitive conception). The teacher and students in this drawing are participating in activities of manipulating testing, exploring, observing, or sensemaking (implementation and social activities). These teacher and these students are participating in scientific activities and learning practices with others using scientific language

and using scientific tools (cognitive, social conceptions).



Figure 16. An example of the highest score of a participant's drawing of conceptions of science instruction.

Participants' Conceptions of Science Instruction: Narratives

The analysis of participants' narratives revealed that eight conceptions of science

instruction for teaching children in pre-K to 3rd grade classrooms were prevalent among

participants in this study. These conceptions included:

- 1. Science instruction as teachers demonstrating science concepts (Demonstration 61%)
 - 1.1 Science instruction as a demonstration (Passive Learning 40%)
 - Demonstration

- Demonstration + Explaining
- Demonstration + Enjoyment
- Demonstration + Observation
- Demonstration + Safety
- Demonstration + Safety + Enjoyment
- Demonstration + Experiment + Engagement
- 1.2 Science instruction as a demonstration (Active Learning 21%)
 - Demonstrations + Inquiry
 - Demonstrations + Engagement
 - Demonstrations + Engagement + Interaction
 - Demonstrations + Engagement + Inquiry
 - Demonstrations + Inquiry + Safety
 - Demonstrations + Inquiry + Engagement
 - Demonstrations + Inquiry + Observation
 - Demonstrations + Inquiry + Experiment + Engagement + Enjoyment
 - Demonstrations + Observation + Inquiry
 - Demonstrations + Observation + Experiment + Interaction + Safety
 - Demonstrations + Experiment + Enjoyment + Inquiry
 - Demonstrations + Enjoyment + Experiment
- 2. Science instruction as making the lesson engaging (Engagement 9%)
 - 2.1 Science instruction as making the lesson engaging (Active Learning 7%)
 - Engagement + Inquiry
 - Engagement + Inquiry + Interaction
 - Engagement + Inquiry + Safety
 - Engagement + Inquiry + Experiment
 - Engagement + Inquiry + Experiment +Interaction
 - Engagement + Inquiry + Enjoyment +Experiment
 - 2.2 Science instruction as making the lesson engaging (*Passive Learning 2%*)
 - Engagement
 - Engagement + Interaction + Enjoyment

- 3. Science instruction as teachers and students engaging in science experiments (*Experiment 9%*) (*Active Learning*)
 - Experiment + Demonstration + Safety
 - Experiment + Inquiry
 - Experiment + Inquiry + Interaction + Engagement
 - Experiment + Inquiry + Observation + Interaction + Engagement
 - Experiment + Inquiry + Engagement + Enjoyment
 - Experiment + Inquiry + Interaction + Observations
 - Experiment + Observation
- 4. Science instruction as implementing science safety guidelines (Safety 8%) (Active Learning)
 - Safety
 - Safety + Demonstrations
 - Safety + Demonstration +Enjoyment
 - Safety + Enjoyment + Interaction
 - Safety + Experiment + Interaction + Engagement
 - Safety + Interaction
 - Safety + Interaction + Demonstration + Enjoyment
- 5. Science instruction as being inquiry-based (Inquiry 8%) (Active Learning)
 - Inquiry
 - Inquiry + Engagement
 - Inquiry + Enjoyment
- 6. Science instruction as students observing scientific phenomena (*Observations 2%*) (*Active Learning*)
 - Observations
 - Observation + Demonstration + Experiment + Engagement
- 7. Science instruction as making the lesson joyful (Enjoyment 2%) (Passive Learning)
 - Enjoyment + Experiment
 - Enjoyment + Safety + Observation
- 8. Science instruction as students interacting with one another and/or with the teacher in science activities (*Interaction 1%*) (*Active Learning*)
 - Interaction + Demonstrations

For clarity, I next define the key constructs, *Demonstration*, *Engagement*, *Inquiry*, etc. inherent in each of the eight conceptions of science instruction. *Demonstration*, according to participants, is an activity to passively and/or actively show science concepts/principles to students. *Engagement*, according to participants, is an effective way to show teacher enthusiasm, spark students' interest, and initiate scientific inquiry, and to get students to be involved with science concepts/principles. The term engagement also refers to the degree of attention, curiosity, interest, optimism, and passion that students show when they are learning or being taught science. Experiment, according to participants, involves an experimental set-up where their students, and they themselves are engaged in scientific practices for doing science. Safety, according to participants is the notion of modeling and instilling the skills and knowledge of science safety to students. *Inquiry*, according to participants, is a strategy whereby students and participants become engaged in cognitive and procedural activities. Cognitive activities involved students' problem-solving through the construction of questions, hypothesis testing, and formulation of solutions/tentative answers for scientific phenomena. Procedural activities involved students' engaging in experiments, and collecting and analyzing data to provide evidence for or against their hypotheses. Observation, according to participants, involved providing their students with scientific practices and tools to closely examine scientific phenomena. Enjoyment according to participants, is the notion of using fun, the "wow" factor, and interest to encourage the learning of science content. Interaction, according to participants is teachers' getting their students and themselves involved in learning science concepts/principles through teacher-students participation structures and through

67

group work. In the following sections, a detailed account of each conception of science instruction is presented.

Demonstrations

Analysis of participants' narratives indicated that science instruction as demonstrations predominantly involved participants' engaging in a number of actions leading to students' passive learning of science content (40 out of 61 = 66%), and to students' active learning of science content (21 out of 61 = 34%) respectively (see Table 4). Table 4 details the numbers and percentages of participants' conceptions of science instruction as demonstrations for passive learning and for active learning and includes the demonstrations associated with other conceptions such as, *Enjoyment, Interaction, Inquiry, Experiment,* and *Safety*. Figure 17 and Figure 18 are pie-charts that illustrate the percentages of participants' conceptions of science instruction as a demonstration, underscored by students' passive learning and underscored by students' active learning, respectively.

Table 4

Numbers and Percentages of Participants' Conceptions Science Instruction as Demonstrations
for Passive Learning and for Active Learning

		Ν	%
	Science instruction as a demonstration (Passive Learning)		
1.	Demonstration	15	
2.	Demonstration + Explaining	8	
3.	Demonstration + Enjoyment	6	
4.	Demonstration + Observation	4	
5.	(i) Demonstration + Safety	4	
(ii) Demonstration + Safety + Enjoyment		2	
6.	Demonstration + Experiment + Engagement	1	
Total		40	66

	N	%
Science instruction as a demonstration (Active Learning)		
1. (i) Demonstrations + Inquiry	6	
(ii)Demonstrations + Inquiry + Safety	2	
(iii)Demonstrations + Inquiry + Engagement	1	
(iv) Demonstrations + Inquiry + Observation	1	
(v)Demonstrations + Inquiry + Experiment + Engagement + Enjoyment	1	
2. (i) Demonstrations + Engagement	3	
(ii) Demonstrations + Engagement + Interaction	1	
(iii) Demonstrations + Engagement + Inquiry		
3. (i) Demonstrations + Observation + Inquiry	1	
(ii)Demonstrations + Observation + Experiment + Interaction + Safety		
4. Demonstrations + Experiment + Enjoyment + Inquiry	1	
5. Demonstrations + Enjoyment + Experiment	1	
Total	21	34



Figure 17. The diagram indicates the percentages of participants' narrative conception of science instruction as a type of demonstration (passive learning).



Figure 18. The diagram indicates number of percentages of participants' narrative conception of science instruction as a type of demonstration (active learning).

Analysis of data indicated that participants who conceptualized teaching as using

demonstrations for passive learning of science content used posters, drawings, diagrams,

models, textbooks, pictures, etc., to convey the science content to their students. Table 5

provides excerpts from participants' narratives that provide evidence of participants'

conceptions of science instruction as demonstrations for passive learning of science content.

Table 5

Examples of Participants' Conceptions of Science Instruction as Demonstrations for Passive Learning

Participant	Demonstrations	Science Content
Susan	Here you can see the teacher teaching about our solar system. She has put the planets in order from closest to farthest from the sun. Once this is done she explains how the planets orbit the sun and this is due to the suns gravitational pull on the planets. Afterwards she tells the students the differences between the planets' orbit and how different	Solar System

Participant	Demonstrations	Science Content
	they are. In the end the students should understand all of the different components of our solar system.	
	Demonstrations + Explaining	
Olive	This teacher is presenting/introducing the water cycle to her class. She defines the terms "evaporation", "condensation", and, "precipitation", to her students and provides visuals to help explain better. This "water cycle" flow chart is drawn on a white board and has the terms written on the side, using arrows to show the steps and flow of the cycle. At first she draws the cycle and has students describe what they see. Then she goes into details and explains what is actually happening.	Water Cycle
	Demonstrations + Enjoyment	
Vivian	Mr. Wright is teaching second graders about the scientific method. The children are sitting at the science lab tables and listening quietly. Mr. Wright is going over the steps of the scientific method while the children quietly repeat after him. The children are having fun while Mr. Wright is teaching. Mr. Wright is enjoying teaching.	Science Process Skills
γοί	This is a kindergarten science class. The teacher is demonstrating the growth cycle of a bean plant. The teacher has brought in all of the necessary supplies: dirt, water, seeds, and containers. The process is demonstrated step by step. The children observe the results of the processes	Life Cycle (Plants)
	Demonstration + Safety	
Ruth	Children are learning about the life cycle of a plant and how it utilizes the process of photosynthesis. The teacher utilizes flowers and add dirt to the soil to demonstrate how water is absorbed through the roots. She wears safety equipment to teach early on safety guidelines. She engages the children by allowing them to water the plants. She speaks throughout the whole process explaining everything.	Photosynthesis Life Cycle (Plant)
	Demonstration + Safety+ Enjoyment	
Eliza	The science teacher has a demonstration after a full day of teaching about volcanoes to give kids a visual of how volcano's work. She follows all safety guidelines and her kids stay in seats. She has them stay behind the safety lines and wear safety goggles. She has them wear gloves and lab coat to show safety. All kids love it but some are lost but are intrigued.	Volcanoes
	Demonstration + Experiment + Engagement	
Piper	The teacher I have drawn is instructing the class on how to conduct an experiment with non-newtonian fluid. To create the fluid, she has added one part water to two parts corn start. Once the fluid is made she will pass it around to the students and have them touch it. First they will try punching the fluid with force and it will remain a solid. Then they will gently push their fingers into it and it seems as if it is a liquid. The teacher will then explain why the fluid does this.	States of Matter

As evident from the data, participants' demonstrations for passive learning of science content included:

1. Only demonstrations that were used by participants to show science content to their students. Participants who conceptualized science instruction as only demonstrations just showed the science content to their students using posters, textbooks, chalkboard, scientific models, pictures, etc. to convey the science content to their students. Within these participant narratives, students did not interact with their peers and teachers but were required only to listen and pay attention to the participants and the science content that was showcased through the demonstrations.

2. Demonstrations used by participants to show science content with participants explaining the science content to their students using the demonstrations. Within these narratives, participants specifically used posters, textbooks, chalkboard, scientific models, pictures, etc. but used the aforementioned to explain the science content to their students. Thus, this was different from the conception of science instruction as only demonstrations wherein participants just showed the science content but without a component of teacher explanation.

3. Demonstrations that showed science content for the underlying purpose of student enjoyment. Within these narratives, participants' mentioned how they used a drawing, a poster, and procedural steps but connected these uses with students' enjoyment instead of students' learning of science content. The use of demonstrations created learning environments that were fun.

4. Demonstrations used by participants to show science content that specifically made

72

provisions for students' observation of scientific phenomena. Within these narratives, participants' mentioned that students had the opportunity to observe scientific activities with a teacher explanation component.

5. Demonstrations used by participants (i) to model science safety to students, and (ii) to model science safety to students for the underlying purpose of students' enjoyment. Within these narratives, participants' mentioned the wearing of googles and lab coats to demonstrate the issue of school science safety. In addition, the goal was also to socialize students in to school science safety. In some cases an added conception of enjoyment was added to make the socialization fun for the students.

6. Demonstrations used by participants that led to students doing an experiment with student engagement. Within these narratives, participants' mentioned how to conduct an experiment and how the experiment would keep students interested and focused on the science content that the experiment was demonstrating.

Analysis of data indicated that participants who conceptualized teaching as using demonstrations for active learning of science content also used posters, drawing, diagrams, models, textbooks, pictures, etc., to convey the science content to their students. This conceptualization was different from that of demonstrations as passive learning of science content because active learning involved students being cognitively engaged with the demonstrations (see Table 6).

73

Table 6

Examples of Participants' Science Instruction as Demonstrations for Active Learning

Participant	Demonstrations	Science Content	
	Demonstrations + Inquiry		
Leah	The teacher is demonstrating a simple baking soda & vinegar & is using parent volunteers to help each group of kids participate hands on. Students would have gloves & cover ups when participating & record information in prepared lab manuals to teach the scientific process.	Chemical Reaction	
	Demonstration + Inquiry + Safety		
Ada	Because this is a science class in an early childhood classroom, teacher is teaching the simple daily things in class. Such as leaves or flowers. This class can be outside the classroom, such as in the playground or garden, to show the students actual plants and also can ask the students to find and name the plants themselves. The class should be in school to make sure students are safe.	Plants	
	Demonstration + Inquiry + Engagement		
Talia	A teacher showing her students how to melt chocolate naturally. The piece of chocolate is solid at the table. She takes the chocolate outside, and she uses a stopwatch to see how long it takes for the sun to melt it. When the chocolate is fully melted, she takes it inside and the students are able to identify how long it took to melt. She then lets the students have their own chocolate and allows them to eat it.	States of Matter	
	Demonstrations + Engagement + Interaction		
Zoe	The teacher in my drawing is holding a test tube with some vinegar in it. In her lesson she is teaching she wants to teach her students the difference between a beakers, test tube graduated cylinder, etc. There are various liquids inside the tubes. She wants the students to communicate the differences/compare/contrast all of the liquids. She tells the students what is in each glass tube. The students have to work in groups.	Science Process Skills	
	Demonstrations + Engagement + Inquiry		
Remi	The teacher will let students hold equipment if appropriate for their age. Science should be a hands-on subject especially for grades 1-3. The teacher should explain what each instrument does and let the child hold it themselves. Teacher should allow students to ask questions about the topic they are learning. Teacher should address each question that the students ask.	Science Process Skills	
Demonstration + Inquiry + Observation			
Paris	In this drawing, the teacher is teaching about the rock cycle. She has 3 different types of rocks as well as a diagram of the cycle. She is going to pass around the rocks, so the students can observe the differences of the rocks. She then will describe the rock cycle. The students will also have the rock cycle diagram to look at.	Rock Cycle	

Participant	Demonstrations	Science Content	
Demonstration + Inquiry + Experiment + Engagement + Enjoyment			
Elena	The teacher is teaching the students about photosynthesis w/a presentation and hands on experiments. By having a presentation and a hands-on activity it allows the students to be involved and have fun while learning	Photosynthesis	
	Demonstration + Observation + Inquiry		
Josie	One product used is a microscope to observe a bug on a flower. Another is a pen and paper to draw an observation. The teacher is explaining how a bird lives in the tree. The students observe how the sun provides life. Students and teacher are exploring different bugs and how they connect with everything around them.	Environmental Science	
Demonstration + Observation + Experiment + Interaction + Safety			
Layla	The teacher is standing at the table as she demonstrates how to use a graduated cylinder, a scale and a test tube as the students are watching. The tube actually has a substance in it and the teacher is demonstrating an experiment. The board has some of the rules that the students should follow and know when participating in an experiment.	Science Process Skills	
Demonstration + Experiment + Enjoyment + Inquiry			
Норе	I drew a teacher showing her students a science experiment. The students I drew look happy. I drew one student looking shocked because the experiment is really cool. The teacher is doing an experiment because science should be hands on. The kids love this rather than reading.	s No identifiable science content	
	Demonstrations + Enjoyment + Experiment		
Rose	The teacher is acting as a scientist in a science lesson. As a class everyone is going through and learning the scientific method as they conduct an experiment. In the picture they are finishing the experiment and the children are engaged and excited and amazed at what is happening.	Science Process Skills	

Participants' demonstrations for active learning of science content included five

variations each centered on demonstrations.

1. In most cases, demonstrations were paired up with inquiry (13/21), and in these

cases demonstrations were used by participants to show science content that then led to

students being engaged with an inquiry activity using the science content from the

demonstration. Variations of this conception included these conceptions also associated with

safety, engagement, observation, experiment, and enjoyment.

2. Demonstrations used by participants to actively engage students with science also included demonstrations that were associated with the conception of engagement, interaction, and inquiry. Demonstrations used by participants to actively engage students with science through inquiry involved getting students interested in the science content first; this was then followed by students' being engaged in cognitive and procedural activities.

3. Apart from the demonstration and inquiry there were two variations whereby demonstrations were paired with observations: in one of these demonstrations were paired-with observations leading to an inquiry activity while the other was followed up with an experiment that included interaction and science safety activities.

4. Demonstrations used by participants were followed up by an experiment underscored by the conception of enjoyment, and this finally led to an inquiry activity.

5. Demonstrations used by participants and underscored by the conception of enjoyment included a student led experiment.

Engagement

For participants in this study, the conception of science instruction as engagement also took on an active and a passive learning component. Analysis of data revealed that for one participant the conception of science instruction as engagement as passive learning involved the conception of simply getting their students interested and motivated towards learning the science content using a teacher-centered activity. For this participant, the conception of science instruction as engagement did not include other conceptions like inquiry, interaction, etc. Table 7 details the numbers and percentages of types of participants 'conceptions of science instruction as engagement.

76

Table 7

Numbers and Percentages of Participants' Conceptions of Science Instruction as Engagement

Science instruction as engagement	Ν	%
Engagement (Passive learning)	1	
Engagement + Inquiry	2	
Engagement + Inquiry + Interaction	1	
Engagement + Inquiry + Safety	1	
Engagement + Inquiry + Experiment	1	_
Engagement + Inquiry + Experiment +Interaction	1	
Engagement + Inquiry + Enjoyment +Experiment	1	_
Engagement + Interaction + Enjoyment (Passive learning)	1	
Total	9	9%

Seven out of the nine participants included the conception of inquiry within their conception of science instruction as engagement. That is, engagement was integrated with both cognitive and procedural activities which led to either interaction, science safety, an experiment, an experiment and interaction or activity underscored by enjoyment but including an experiment component. Figure 19 presents a pie-chart that illustrates the variations and percentage of participants' narrative conceptions of science instruction as engagement. Table 8 provides excerpts from participants' narratives that were predominantly associated with the conception of science instruction as engagement.



Figure 19. Diagram of the percentages of participants' narrative conceptions science instruction as *engagement* (active learning).

Table 8

Excerpts from Participants' Narratives that were Predominantly Associated with the Conception of Science Instruction as Engagement

Participants		Science Content
	Engagement (Passive Learning)	
Ellie	I would give each child a flower pot, seeds and potting soil. We would plant the seeds ad begin watering them and exposing them to sunlight. I would explain to the students the process of photosynthesis and why the plants are growing in terms that a 3 rd grader would understand. I would test their knowledge of the process. I would send the plants home with the students along	Photosynthesis

Participants		Science Content
	with the information on what we have learned about photosynthesis.	
	Engagement + Inquiry	
Ava	The teacher is walking around as students are engaged in activities. She is facilitating learning. In a science classroom, I anticipate it being very hands-on. I want children to touch things, explore, ask questions, think about things, and much more. I envision the science classroom as a place where children can explore our world.	Science Process Skills
	Engagement + Inquiry + Interaction	
Zоеу	There is a science sensory/hands on center in the classroom. Teacher lets the children explore/lead. There is more action learning than lecture learning. Teacher is doing hands on science activities with the children. Teacher follows and integrates science TEKS in the Kindergarten classroom.	Science Process Skills
	Engagement + Inquiry + Safety	
Stella	This teacher as scientist is allowing the students to learn in hands on ways outdoors. Getting the children up and moving engages the students. In this picture, the students are learning about life cycles such as that of a butterfly. They see where a caterpillar may live, what it eats and will get to watch it go through its cycle throughout the weeks (the teacher will have a safe place for them to grow). Once is has reached its final stage, the children get to release the now butterflies.	Life Cycle (Animals)
	Engagement + Inquiry + Experiment	
Luna	The teacher is teaching a science lesson and the students have supplies on their desks. Science is very hands on so rather than just showing, the teacher is letting the students do the experiment alongside the teacher.	No identifiable Science Content
	Engagement + Inquiry + Experiment + Interaction	
Alice	A science teacher should facilitate an active learning environment. Allowing hands on learning over a traditional lecture. The teacher should teach thru experimentation and allow students to make assumptions they can test. The classroom should be inviting so all students can get involved. Students should also have the options to work in a group or individually	No identifiable Science Content
	Engagement + Inquiry + Enjoyment + Experiment Content	
Iris	A good science teacher in the early childhood classroom participates with her students. She is on their level and is	Physical Science (Simple Machines)

Participants		Science Content
	guiding their learning while they get to be hands on and have a full experience. In this example, the teacher us leading an activity where students explore physical science by seeing what rolls sown an incline. Each get to be hands on and have their own trial. There is also a literacy component. It is fun and promotes social science, and literacy skills.	
	Engagement + Interaction + Enjoyment (Passive Learning)	
Mila	This is a teacher teaching her students environmental science. Her kids are sitting on the rug while learning about recycling. The teacher is normally dressed. She will give them an interactive assignment. The kids will find it fun.	Environmental Science

For these four participants, teacher-centered activities involved and connected students with the science content through inquiry activities such as, "touching things, exploring, asking questions, thinking about things," "exploring/leading," "learning," and making "assumptions they can test." Apart from the conceptions of engagement and inquiry, some participants connected the aforementioned conceptions with conceptions like interaction, safety, experiment, and experiment and interaction. For example, Zoey's narrative indicated that she connected the conceptions of engagement and inquiry with the conception of interaction -"The teacher is doing hands on science activities with the children" while Stella connected conceptions of engagement and inquiry with the conception of safety - "...the teacher will have a safe place for them to grow." For Luna, the conception of science instruction as engagement and inquiry was connected with the conception of experiment – "the teacher is letting the students do the experiment alongside the teacher." For Alice, conceptions of engagement and inquiry were connected the with conception of experiment " ... the teacher should teach through experimentation" - and with the conception of interaction – " Students should also have the options to work in a group or individually." Lastly, among the nine participants, only

Mila, who also centered her conception of science instruction as engagement, made a

connection with the conceptions of interaction and enjoyment. For Mila interaction was her use

of an "interactive assignment" which together with engagement let the students in her class

have "fun."

Experiment

Analysis of narrative data indicated that for those participants who conceptualized

science instruction as doing experiments, the conception of doing an experiment was

associated predominantly with the conception of inquiry and to a lesser extent with the

conceptions of demonstration and observation (see Table 9).

Table 9

Number of Participants' Narrative Conceptions of Science Instruction as Experiment (involving Active Learning)

Science instruction as experiment	Ν	%
Experiment + Demonstration + Safety	1	
Experiment + Inquiry	1	
Experiment + Inquiry + Interaction + Engagement	1	
Experiment + Inquiry + Observation + Interaction + Engagement	1	
Experiment + Inquiry + Engagement + Enjoyment		
Experiment + Inquiry + Interaction + Observations	1	
Experiment + Observation		
Total	9	9

As is evident from Table 9, one variation of science instruction as doing experiments was associated with the conception of demonstration and safety. Science instruction as doing experiments was also associated with the conception of inquiry, and included the conceptions of experiments and inquiry with (a) the conceptions of interaction and engagement, (b) the conceptions of observation, interaction, and engagement, (c) the conceptions of observation, engagement, and enjoyment, and (d) the conceptions of interaction and observations. Lastly, science instruction as doing experiments was associated with the conception of demonstration.

Figure 20 presents a pie-chart that illustrates the variations and percentages of participants' narrative conception of science instruction as Experiment. Table 10 provides excerpts from participants' narratives that were predominantly associated with the conception of science instruction as experiment.



Figure 20. Diagram of the percentages of participants' narrative conceptions of science instruction as experiment (in active learning).

Table 10

Participants' Narratives Indicating Conceptions of Science Instruction as Experiment (in Active Learning)

Participants		Science Content
Experiment + De	emonstration + Safety	
Ember	Mrs. Sunshine is teaching the kids a new scientist experiment on science Monday. Last week the students learned about the reaction between Baking soda and vinegar. Mrs. Sunshine wanted to keep the same theme going. Mrs. Sunshine decided to show these kids the reaction between dry ice and water. Following all safety rules she was able to successfully complete the experiment.	Chemical Reaction
Experiment + Inc	quiry	
Tessa	The teacher is going to allow the students to experiment in the classroom. Students will share a variety of different objects. They will test to see which sink and which float.	Sink and Float
Experiment + Inc	quiry + Interaction + Engagement + Enjoyment	
Melody	The scientist is professionally dressed. They are doing an experiment for the classroom It is exciting experiment to entice the kids. They are happy because they are having fun. It is a student-centered activity.	No Identifiable Science Content
Experiment + Inc	quiry + Observation + Interaction + Engagement	
Ruby	In both pictures, students are learning about and exploring measurements of liquids. They are given the opportunity to observe the teacher, discuss and ask questions. They are also given the opportunity to explore hands-on by measuring on their own while the teacher observes/assists as needed. They are measuring using beakers and graduated cylinders. This is a more interactive, hands- on science lesson.	Science Process Skills
Experiment + Inc	quiry + Engagement + Enjoyment	
Ariel	In the drawing, a student is pouring a liquid into a petri dish. The other students are then creating a hypothesis based on the experiment. The teacher is observing the children conducting the experiment and guiding them. The teacher is there to help the kids have a fun and safe time.	Science Process Skills
Experiment + Inquiry + Interaction + Observations		
Freya	In this classroom, the students are conducting a hands-on lab where each day they get to observe the life cycle of a butterfly. They work in groups of 2-3 to decrease the number of insects needed as well as establish teamwork. Needed for the laboratory is a certain number of butterflies, plants, and insect cages and food for the animal to survive. Objects for learning, release them, and observe in their natural habitat. During the study, students will write down their observations and thoughts.	Life Cycle (Animals)

Participants		Science Content
Experiment + C	Observation	
Nicole	In a classroom with such young learners, I decided to pick a science experiment that they would understand. The process involves 3 white flowers (of any kind), 3 vases, and 3 food coloring of your choosing. To complete this experiment, you would cut the bottom of the stem of each rose and place them into vases with water that has been dyed. As the water moves up the rose, the students can see where it travels. This activity allows students to see how water travels up plants and almost "defies gravity".	Transpiration

In most of these five narratives, the experiment was underscored by a science content and the inquiry activity involved students engaged in a minds-on activity - "doing", "discussing and asking questions, ""measuring", "creating a hypothesis," "conducting a hands-on lab" and writing down "observations and thoughts." Apart from this, participants like Melody mentioned that her students would interact and engage with the content through a student-centered activity, while Ruby included the conceptions of observation, interaction, and engagement whereby her students observed their teacher modeling the science content and then interacted and engaged with the science content and herself as the teacher – "They are also given the opportunity to explore hands-on by measuring on their own while the teacher observes/assists as needed. They are measuring using beakers and graduated cylinders. This is a more interactive, hands-on science lesson." For Ariel, the conceptions of experiment and inquiry were connected more with the conceptions of engagement - "...a student is pouring a liquid into a petri dish ... students are then creating a hypothesis based on the experiment" – and enjoyment - "kids have a fun and safe time." For Freya, the conceptions of experiment and inquiry were connected more with the conceptions of interaction – "They work in groups of 2-3 to decrease the number of insects needed as well as establish teamwork" - and observation " ... observe in their natural habitat. During the study, students will write down their observations and thoughts."

Among the participants who showed conceptions of science instruction as doing experiments, there was one participant, Ember, for whom the conception of experiment was more clearly one of her doing the experiment to show the idea of a chemical reaction and keeping students aware of the safety rules for that particular experiment. Lastly, three participants, Hazel, Diana, and Nicole conceptualized science instruction as themselves doing experiments – "science experiment," "For the experiment, the teacher puts the plant near the window" and "I decided to pick a science experiment … "…with their students observing the experiments and learning the science content from noticing, "watching", and seeing the experiment.

Safety

Analysis of narrative data indicated that for those participants who conceptualized science instruction as implementing science safety guidelines, the conception of implementing science safety was predominantly associated with the conception of demonstrations, and the conception of demonstration led to students' enjoyment in science safety activities. The conceptions of implementing science safety was associated with the conception of enjoyment, experiment, and interaction (See Table 11). Figure 21 presents a pie-chart that illustrates the variations and percentage of participants' narrative conception of science instruction as Implementing Science Safety Guidelines. Table 12 provides excerpts from participants' narratives that were predominantly associated with the conception of science instruction as Implementing Science Safety Guidelines (active learning).

85

Table 11

Number of Participants' Narrative Conceptions of Science Instruction as Implementing Science Safety Guidelines (Active Learning)

Science instruction as implementing science safety guidelines (Safety)	Ν	%
Safety	2	
Safety + Demonstrations	1	
Safety + Demonstration +Enjoyment	1	
Safety + Enjoyment + Interaction	1	
Safety + Experiment + Interaction + Engagement	1	
Safety + Interaction	1	
Safety + Interaction + Demonstration + Enjoyment	1	
Total	8	8



Figure 21. Diagram of the percentages of participants' narrative demonstrating conceptions of science instruction as implementing science safety guidelines (in active learning). Table 12

Participants' Narratives Indicating Conceptions of Science Instruction as Implementing Science Safety Guidelines (in Active Learning)

Participants		Science Content
	Safety	
Sadie	The teacher is teaching the life cycle in a 1 st grade classroom. She is wearing goggles because it is important to be safe during labs. I	Life Cycles

Participants		Science Content
	interned in a 1 st grade class one year and I remember them learning about life cycles.	
	Safety + Demonstration	
Lucia	Drew the scientist with a lab coat and play glasses for the kids. Put a warning sign for the younger children to be able to see. He is in the middle of the classroom when he introduces the class to him. There are photos behind him to show the class what they will be doing today.	No identifiable Science Content
	Safety + Demonstration +Enjoyment	
Thea	A teacher is wearing goggles and a lab coat. She is holding up a beaker, showing a chemical reaction. In the other hand, she is displaying pictures of leaves. She is very happy. The children are excited too.	Chemical reaction
	Safety + Enjoyment + Interaction	
Violet	The teacher is making sure that the students are learning in a safe environment, and making their learning fun and enjoyable. Having hands on activities and lessons allow for the student to retain the information that the teacher is wanting the students to know. A good science teacher makes it very fun and informative!	No identifiable Science Content
	Safety + Experiment + Interaction + Engagement	
Sophia	The teacher and students are in lab coats and safety goggles to follow safety procedures during the experiment. The teacher is allowing the students to interact with the baking soda volcano. She has allowed one students to put the baking soda in the volcano, then she allowed the other student to put in the vinegar. By allowing the students to interact with the experiment, she is encouraging them to actively learn in class. She also had the girls put their hair up to promote proper safety practices.	Chemical reaction
	Safety + Interaction	
Cali	The teacher and students wear googles and lab coat. The students are sitting at tables big enough to work safely. On the tables are work stations for the students.	No identifiable Science Content
	Safety + Interaction + Demonstration + Enjoyment	
Aria	The teacher is wear a short sleeve lab coat. The kids are playing with lab equipment. The teacher is describing to the class how to use a graduated cylinder. The kids have to wear closed toe shoes. She is keeping the kids interested in the lesson.	No identifiable Science Content

For participants in this study, the conception of science instruction as implementing

safety in the pre-K to-3rd grade classroom was centered on both the participants and their

students following safety rules/procedures. These safety rules predominantly included both

participants and students wearing lab coats and googles, and to a lesser extent having "girls put their hair up to promote proper safety practices," or having kids "wear closed toe shoes" or having a "safe environment."

Inquiry

Analysis of narrative data indicated that for those participants who conceptualized of science instruction as being inquiry-based, participants' narratives conceptualizations of science instruction as being inquiry were predominantly associated with the conception of *engagement and enjoyment* (see Table 13). Table 13 indicates that participants' narratives conceptions of science instruction as being inquiry-based were predominantly associated with the conception of engagement, and the conception of enjoyment. Figure 22 presents a pie-chart that illustrates the variations and percentages of participants' narrative conceptions of science instruction as being inquiry-based sciences from participants' narratives that predominantly indicated conceptualizations of science instruction as being inquiry based.

Table 13

Number of Participants' Narrative Conceptions of Science Instruction as Being Inquiry-Based (Inquiry) (in Active Learning)

Science instruction as being inquiry-based (Inquiry)	Ν	%
Inquiry	5	
Inquiry + Engagement	2	
Inquiry + Enjoyment	1	
Total	8	8



Figure 22. Diagram of the percentages of participants' narrative conceptions of science instruction as being inquiry-based (*Inquiry*) (in active learning).

Table 14

Excerpts from Participants' Narratives Conceptions of Science Instruction as Being Inquiry Based (Inquiry)

Participant		Science Content
	Inquiry	
Grace	When I think of scientist in an early childhood classroom, I think about how my teachers incorporated the outdoors. We learned about growing plants and used aprons and watering cans when we worked with the plants.	Growth (Plants)
Kate	The drawing show 3 different soils, one is just dirt, the other has a little flower, and the third one has a couple of flowers grew in there. They are all in one table so the students can move around and take a look at them. The activity is hand-on because early- childhood students tend to learn better this way. The teacher is asking question to activate their critical thinking. They are in	Life Cycle (Plants)

Participant		Science Content
	groups because if someone does doesn't understand, or the teacher instruction are not clear to them, they can always turn to a classmate who can help.	
Vera	The teacher is teaching the class the life cycle of plants. A student is raising his/her hand to ask a question. The class size is small, an ideal concept to follow.	Life Cycle (Plants)
Claire	The teacher has passed around planet models in the form of a ball so that they can hold them and participate in kinesthetic learning.	Solar System
Lucy	I put flasks and beakers on the table because children learn best when working with tangible products. I put the teacher in front of the class that way he able to be heard by all the students. There is writing on the board that way students are able to refer back to the board if they forget something. If a student has a question, they are expected to raise their hand.	Science Process Skills
	Inquiry + Engagement	
Maria	Science in an early childhood classroom is about curiosity. In this picture you see the teacher leading the students in a "sink or float" activity lesson. Children are allowed to touch the objects and place in the bucket of water on the table. The teacher then asks and writes the outcomes on the chart, to provide literacy.	Sinking and Floating
	Inquiry + Enjoyment	
River	Science should be full of hands-on activities so that it's fun for kids. If it's fun for the kids, there's a good chance that they will remember the information. Science tends to be dreaded by some students, so we should make it as painless as possible.	No Identifiable Science Content

The conception of science instruction as being inquiry-based, i.e. of science activities as

having both hands-on components and minds-on (cognitive activities) components, was

mentioned by only eight participants. Among these eight participants, science instruction as

being inquiry-based was associated with a specific science content. River was the only

participant who did not mention any science content, but she mentioned that the hands-on

activities she planned to use would help her future students to remember "the information."

Participants who associated their conception of science instruction as inquiry-based with

science content mentioned a hands-on activity that was then connected to a minds-on activity. Kate's narrative was exemplary of the association between inquiry and science content (see Table 14). In most cases, the minds-on activity was focused on students "asking questions," participating in "kinesthetic learning," "making note to refer back," writing outcomes, "taking notes," creating a "flip book," etc. Additionally, the conception of science instruction as being inquiry-based showed associations with other conceptions. For example, Maria connected her conception of science instruction as being inquiry-based with the conception of engagement – "Children are allowed to touch the objects and place them in the bucket of water on the table" -while River connected her conception of science instruction as being inquiry-based with the conception of engagement.

Observations

Analysis of narrative data indicated that for those participants who conceptualized science instruction as observation, the conception of observation was predominantly associated with the conception of demonstrations leading to students' experimentation and engagement in observation activities. (See Table 15). Figure 23 presents a pie-chart that illustrates the variations and percentages of participants' narrative conceptions of science instruction as *Observation*. Table 16 provides excerpts from participants' narratives that were predominantly associated with the conception of science instruction as *observations*.

Table 15

Number of Participants' Narrative Conceptions of Science Instruction as Students Observing Scientific Phenomena (Observations)

Science instruction as students observing scientific phenomena	N	0/
(Observations).	IN	/0

Observations	1	
Observation + Demonstration + Experiment + Engagement	1	
Total	2	2



Figure 23. Diagram of the percentages of participants' narrative conceptions of science instruction as students observing scientific phenomena (*observations*) (in active learning).

Table 16

Excerpts from Participants' Narratives Conceptions of Science Instruction as Observations

Participant		Science Content
	Observation	
Cora	The teacher is teaching her students the process of photosynthesis. The kids are observing the sun giving off light, the flower taking in the light & CO2 and then giving off oxygen. The kids are learning that this is how plants make food and help humans.	Photosynthesis

Participant		Science Content
	Observation + Demonstration + Experiment + Engage	ement
Lillian	The kids are watching her as she demonstrates the experiment. She is making elephant toothpaste. It is engaging while introducing chemistry.	Chemical Reaction

The conception of science instruction as students observing scientific phenomena (Observations) was mentioned by only two participants in the study. For these two participants, science instruction was centered on their students "observing" and "watching" scientific phenomena like photosynthesis and a chemical reaction. For Cora, this conception was focused on the process of photosynthesis while Lillian situated the conception of science instruction as observing scientific phenomena with the conceptions of science instruction as demonstration, experiment, and engagement.

Enjoyment

Analysis of narrative data indicated that for those participants who conceptualized science instruction as enjoyment, this conceptualization was predominantly associated with the conception of experiment, and the conception of science instruction as enjoyment being predominantly associated with the conception of Safety led to the use of observation science activities with students (Table 17). Figure 24 presents a pie-chart that illustrates the variations and percentages of participants' narrative conceptions of science instruction as *enjoyment*. Table 18 provides excerpts from participants' narratives that predominantly conceptualized science instruction as *enjoyment*.

93

Table 17

Number of Participants' Narrative Conceptions of Science Instruction as Enjoyment.

Science instruction as enjoyment	Ν	%
Enjoyment + Experiment	1	
Enjoyment + Safety + Observation	1	
Total	2	2



Figure 24. Diagram of the percentages of participants' narrative conceptions of science instruction as enjoyment (passive learning).

Table 18

Excerpts from Participants' Narratives Conceptions of Science Instruction as Observations

Participant		Science Content			
	Enjoyment + Experiment				
Nina	This is Mr. Scientist in the 1 st grade today! The kids are very excited to be involved in some experiments.	No identifiable Science Content			
Enjoyment + Safety + Observation					
Pearl	The teacher is dressed as a "mad scientist." Everybody (including the children) is wearing goggles. The children are watching while maintaining a safe distance. Everybody has a smile because the teacher is making science fun	No identifiable Science Content			

Analysis of participants' narratives revealed only two participants who conceptualized science instruction as being centered on enjoyment. For these two participants, Nina and Pearl, their conception of science instruction lacked any association with science content. For Nina, the notion of students being "excited" about being involved in some experiments was central to her conception of science instruction. For Pearl, the conception of science instruction as enjoyment translated into an issue of safety and observation of her dressed as a "mad scientist."

Interaction

Analysis of narrative data indicated that for those participants who conceptualized science instruction as interaction, this conceptualization was predominantly associated with the conception of science instruction as demonstrations (see Table 19). Figure 25 presents a piechart that illustrates the percentages of participants' narrative conceptions of science instruction as interaction. Table 20 provides excerpt from participants' narratives that were predominantly associated with the conception of science instruction as being interaction. Table 19

Number of Participants' Narrative Conceptions of Science Instruction as Interaction

Science instruction as being interaction	Ν	%
Interaction + Demonstrations	1	
Total	1	1



Figure 25. Diagram of the percentages of participants' narrative conceptions of science instruction as *interaction* (in active learning).

Table 20

Excerpts from Participants' Narratives Conceptions of Science Instruction as Interaction

Participant	Interaction + Demonstration	Science Content
	The teacher will join in the classroom activities with students. The	
Amira	product is a classroom activity teaching kids about life science. The	Life Science
	teacher uses the model and let kids dance to experience.	

Only one participant's narrative contained the conception of science instruction as

interaction. This interaction involved the teacher joining the students in an activity that used a

model and dancing to convey life science content that was not identified.

Synthesis of Findings from Drawings and Narratives

In this section, I present a synthesis merging of the preliminary themes that arose from

the analysis of drawings, and from the analysis of narratives. The findings from the STLP3,

specifically the dimensions of social, affective, implementation, and cognitive, from the analysis

of drawings were related to the conceptions of science instruction resulting from the analysis of

narratives. Each conception was read and reread in relation to the supporting data evidence

and then correlated with each of four dimensions and with the participants' drawings.

Table 21

Identifying the Numbers of Percentage of Participants' Conceptions of Science Instruction from Participants' Drawings, Participants' Narratives, and Science Content

Conceptions: Drawings	Conceptions: Narratives	Science Content
Social (85%) Participate in scientific activities and learning practices with others, using scientific language and tools. Affective (73%) Experience excitement, interest and motivation to learn about	 Science instruction as a demonstration (Passive Learning) g practices with ng scientific language demonstration (Active Learning) (21%) 	 No identifiable Science Content (18%) Physical Science (19%) Life Science (36%)
	 Science instruction as engagement (Active) (7%) 	 Earth and Space Science (14%)
phenomena in the natural and physical world. Implementation (37%)	 Science instruction as engagement (Passive) (2%) 	Nature of Science (13%)
Manipulate, test, explore, predict, question, observe and make sense of the natural and physical world. Cognitive (33%) Come to generate, understand, remember and use concepts, explanations, arguments, models and facts related to science.	 Science instruction as experiment (Active) (9%) 	
	 Science instruction as being inquiry-based (Active) (8%) 	
	 Science instruction as implementing science safety guidelines (Active) (8%) 	
	 Science instruction as students observing scientific phenomena (Active) (2%) 	
	• Science instruction as being interaction (Active) (1%)	

Table 22

Examples of Merging Themes of Participants' Conceptions of Science Instruction for Children in Pre-K to 3rd Grade Classroom

Conceptions of Science Instruction	Social	Affective	Implementation	Cognitive	
Science instruction as engagement					
Engagement (passive)					
Engagement + Inquiry					
Conceptions of Science Instruction	Social	Affective	Implementation	Cognitive	
--	-----------------	------------	----------------	-----------	--
Engagement + Inquiry + Interaction					
Engagement + Inquiry + Safety					
Engagement + Inquiry + Experiment					
Engagement + Inquiry + Experiment +					
Interaction					
Engagement + Inquiry + Enjoyment +					
Experiment					
Engagement + Interaction + Enjoyment					
(passive)					
Science i	instruction as	inquiry			
Inquiry					
Inquiry + Engagement					
Inquiry + Enjoyment					
Science ins	struction as ir	nteraction			
Interaction + Demonstrations					
Science instruction as observation					
Observations					
Observation + Demonstration + Experiment +					
Engagement					
Science instruction as implementing science safety guidelines (Safety)					
Safety					
Safety + Demonstrations					
Safety + Demonstration + Enjoyment					
Safety + Enjoyment + Interaction					
Safety + Experiment + Interaction +					
Engagement					
Science instruction as a demonstration - Passive Learning					
Demonstration					
Demonstration + Explaining					
Demonstration + Enjoyment					
Demonstration + Observation					
(i) Demonstration + Safety					
(ii) Demonstration + Safety + Enjoyment					
Demonstration + Experiment + Engagement					

Table 23

Comparing the Participants' Narratives Conceptions of Science Instruction for Children in Pre-K to 3rd Grade between Active Learning and Passive Learning

Active Learning	Passive Learning
Participants' conceptions of science instruction as	Participants' conceptions of science instruction
a demonstration with an active learning	as a demonstration with a passive learning
component can be categorized as predominantly	

containing social, implementation, and cognitive component can be categorized as containing dimensions (21%). affective dimensions (40%). Participants' conceptions of science instruction as Participants' conceptions of science instruction engagement with an active learning component as engagement with a passive learning can be categorized as predominantly containing component can be categorized as predominantly social, implementation, and cognitive dimensions containing affective dimensions (2%). (7%). Participants' conceptions of science instruction as Participants' conceptions of science instruction doing experiments can be categorized as as enjoyment can be categorized as predominantly containing social, implementation, predominantly containing social and affective and cognitive dimensions (9%). dimensions (2%). Participants' conceptions of science instruction as implementing safety can be categorized as predominantly containing a social dimension (8%). Participants' conceptions of science instruction as inquiry can be categorized as predominantly containing social, implementation, and cognitive dimensions (8%). Participants' conceptions of science instruction as observation can be categorized as predominantly containing social, implementation, and cognitive dimensions (2%). Participants' conceptions of science instruction as an interaction can be categorized as predominantly containing a social dimension (1%).

As is evident in Table 23, 21% of participants' conceptions of science instruction as a

demonstration within an active learning context can be categorized as predominantly containing social, implementation, and cognitive dimensions. Further, participants' conceptions of science instruction as engagement with an active learning component accounted for 7% and

can be categorized as predominantly containing social, implementation, and cognitive

dimensions. Also, 9% of participants' conceptions of science instruction as doing experiments

can be categorized as predominantly containing social, implementation, and cognitive

dimensions. In addition, participants' conceptions of science instruction as inquiry within an

active learning context accounted for 8% and can be categorized as predominantly containing

social, implementation, and cognitive dimensions. Further, participants' conceptions of science instruction as observation accounted for 2% and can be categorized as predominantly containing social, implementation, and cognitive dimensions. Only one participant's conceptions of science instruction as an interaction can be categorized as predominantly containing a social dimension, and this accounted for 1%.

On the other hand, 40 % of participants' conceptions of science instruction as a demonstration with a passive learning component can be categorized as containing affective dimensions Participants' conceptions of science instruction as engagement with a passive learning component accounted for 2% and can be categorized as predominantly containing affective dimensions. Additionally, participants' conceptions of science instruction as enjoyment accounted for 2% can be categorized as predominantly containing social and affective dimensions.

Summary

Chapter 4 reports the findings from analysis of participants' drawings and participants' narratives of drawings of science instruction for children in pre-K to 3rd grade classrooms. Analysis of participants' drawings using the STLP3 instrument revealed four themes that characterized participants' conceptions of science instruction. These four conceptions were: (a) science instruction in an early childhood classroom is a social activity, (b) science instruction in an early childhood classroom as an affective endeavor, (c) science instruction in an early childhood classroom is the implementation of activities for learning, and (d) science instruction in an early childhood classroom is a cognitive activity. The analysis of participants' narratives revealed that eight conceptions of science instruction for teaching children in pre-K to 3rd grade

classrooms were prevalent among participants in this study. These conceptions included science instruction as teachers demonstrating science concepts (Demonstration) (passive learning and active learning), science instruction as making the lesson engaging (Engagement) (active learning and passive learning), science instruction as teachers and students engaging in science experiments (Experiment) (active learning), science instruction as implementing science safety guidelines (Safety) (active learning), science instruction as inquiry-based (Inquiry) (active learning), science instruction as students observing scientific phenomena (Observations) (active learning), science instruction as making the lesson joyful (Enjoyment) (passive learning), and science instruction as students interacting with one another and/or with the teacher in science activities (Interaction) (active learning). Chapter 5 includes discussion of this study, educational implications, and recommendations for future research.

CHAPTER 5

DISCUSSION

The purpose of this study was to investigate participants' conceptions of science

instruction prior to the commencement of their early childhood education methods courses.

The primary research question for this study was: What are prospective early childhood

teachers' conceptions of science instruction in the early childhood classroom? In summary, the

merging of findings from the analysis of drawings and from the analysis of narratives resulted in

the following major themes:

- 1. Participants' conceptions of science instruction as a demonstration with a passive learning component can be characterized as containing affective dimensions.
- 2. Participants' conceptions of science instruction as a demonstration with an active learning component can be characterized as predominantly containing social, implementation, and cognitive dimensions.
- 3. Participants' conceptions of science instruction as engagement with active learning can be characterized as predominantly containing social, implementation and cognitive dimensions.
- 4. Participants' conceptions of science instruction as engagement with passive learning can be characterized as predominantly containing an affective dimension.
- 5. Participants' conceptions of science instruction as inquiry can be characterized as predominantly containing social, implementation, and cognitive dimensions.
- 6. Participants' conceptions of science instruction as an interaction can be characterized as predominantly containing a social dimension.
- 7. Participants' conceptions of science instruction as observation can be characterized as predominantly containing social, implementation and cognitive dimensions.
- 8. Participants' conceptions of science instruction as implementing safety can be characterized as predominantly containing a social dimension.
- 9. Participants' conceptions of science instruction as doing experiments can be characterized as predominantly containing social, implementation and cognitive dimensions.

10. Participants' conceptions of science instruction as enjoyment can be characterized as predominantly containing social and affective dimensions.

Table 24 summarizes the major themes resulting from the merging of preliminary

themes organized as Active Learning and Passive Learning. In summary, each conception was

anchored by a unique construct (Demonstration, Engagement, Inquiry, Interaction,

Observation, Safety, Experiment, and Enjoyment) and underscored by dimensions (Affective,

Social, Implementation, and Cognitive).

Table 24

Major Themes Resulting from Merging of Findings from Drawings and Narratives

Active Learning		Passive Learning
Social, Implementation and Cognitive Dimensions	Social Dimension	Affective dimension
• Science instruction as a demonstration with an active learning component (21%)	 Science instruction as an interaction (1%) Science instruction as implementing safety (8%) 	 Science instruction as a demonstration with a passive learning component (40%)
 Science instruction as engagement (7%) 		 Science instruction as enjoyment (2%)
• Science instruction as inquiry (8%)		 Science instruction as engagement (2%)
• Science instruction as observation (2%)		
• Science instruction as doing experiments (9%)		

The final major themes of 100 participants' conceptions of science instruction resulting

from the merging of findings from drawings and narratives can be clarified as follows:

1. Science instruction as a demonstration with an active learning component can be characterized as predominantly containing social, implementation, and cognitive dimensions (21%).

- 2. Science instruction as engagement with an active learning component can be characterized as predominantly containing social, implementation, and cognitive dimensions (7%).
- 3. Science instruction as doing experiments with an active learning component can be characterized as predominantly containing social, implementation, and cognitive dimensions (9%).
- 4. Science instruction as inquiry with an active learning component can be characterized as predominantly containing social, implementation, and cognitive dimensions (8%).
- 5. Science instruction as implementing safety with an active learning component can be characterized as predominantly containing a social dimension (8%).
- 6. Science instruction as observation with an active learning component can be characterized as predominantly containing social, implementation, and cognitive dimensions (2%).
- 7. Science instruction as an interaction with an active learning component can be characterized as predominantly containing a social dimension (1%).
- 8. Science instruction as a demonstration with a passive learning component can be characterized as containing affective dimensions (40%).
- 9. Science instruction as engagement with a passive learning component can be characterized as predominantly containing affective dimensions (2%).
- 10. Science instruction as enjoyment can be characterized as predominantly containing social and affective dimensions (2%).

Findings in Relation to Standards

According to the National Association for the Education of Young Children (NAEYC,

2009), the National Science Teacher Association (NSTA, 2014), and the National Research

Council (NRC, 2012), science instruction in early childhood is characterized as ideally taking

place in an inquiry-based and experiential environment, wherein pre-K to 3rd grade children

have the opportunity capacity to engage in scientific practices, and to develop an

understanding of science content and science skills at a conceptual level. This development of

an inquiry-based and experiential environment must also provide pre-K to 3rd grade children with multiple and varied opportunities to engage in science exploration and discovery, thereby creating communities of science learners. Accordingly, these standards also require that teachers create environments in which they and their students work together as active learners, use assessments of students and of their own teaching continually to plan and conduct their teaching, build strong relationships with students, and include their students as active members of science learning communities. The standards provide guidelines for teachers to employ science teaching and learning strategies which enable children in an early childhood classroom to develop science understanding at a conceptual level through *active learning* and *play*.

The findings in this study are partially in line with the aforementioned standards, specifically, participants' conceptions of science instruction that emphasized active learning with social, implementation, and cognitive dimensions. Table 24 indicates that 56% of participants' conceptions of science instruction were underscored by the notion of active learning, in contrast to the 44% of participants' conceptions of science instruction, which were underscored by the notion of passive learning. Even though participants' conceptions of science instruction revealed an active learning orientation, findings also revealed that science instruction as demonstration, as engagement, as inquiry, as observation, and as doing experiments involved superficial understandings of science at a grade appropriate level and at a conceptual level. This finding is also important because it shows the impact of the weak science content preparation and weak preparation in scientific practices of future early childhood

educators and the resulting impact on their conceptions of science instruction. Most importantly, this finding emphasizes the need for early childhood educators to demonstrate a deep understanding of pure and applied science and have the knowledge and skills required to teach students science in age-appropriate, meaningful ways (NSTA, 2017). This finding from this study therefore indicates that participants in this study had limited prior knowledge of developmentally appropriate instructional strategies for children in pre-K to 3rd grade science instruction.

The use of play in developing concepts and the assessment of student learning were not part of participants' conceptions of science instruction. This is in contrast to the literature that calls for providing support for play as a key instructional strategy for promoting young children's development from the infant/toddler stage to the early grades (NAEYC, 2009; NRC, 2012; NSTA, 2014). Thus, early childhood teachers, like the participants in this study, require assistance in the thoughtful planning of the materials and activities that can make possible the use of play to scaffold for science learning in early childhood classrooms. In contrast, the 44% of participants' whose conceptions of science instruction were underscored by the notion of passive learning did conceptualized the use of an affective component for student learning of science but this overshadowed the cognitive aspects of learning science. The use of play in developing concepts, and the assessment of student learning were also not part of participants' conceptions of science instruction that were underscored by the notion of passive learning.

As mentioned, the participants in this study did not hold conceptions that emphasized play as a means of teaching science to young children. This may be surprising in light of the well-researched agenda and literature on play as a crucial element in enabling young children

to build science concepts. Participants in this study and other prospective teachers who are learning to teach science to young children should conceptualize play as an element for the implementation and cognitive dimensions of teaching science. Most importantly, they should not just restrict the notion of play to the affective and social dimensions of teaching science.

Findings in Relation to Current Instructional Strategies: Pre-K to 3rd Grade Science Instruction

In a 2014 position statement, the National Science Teachers Association (NSTA) stressed the need for reflective, inquiry-based science experiences at the early childhood level in order to support and strengthen student learning. The National Research Council recommends the inclusion of inquiry science experiences at the early childhood levels (NRC 2007, 2012). Recommended early science learning experiences related to a guided, inquiry-based approach include the following: identifying and asking questions that can be answered through investigation, designing and conducting investigations, using appropriate tools and equipment, learning to develop logical conclusions, and communicating understandings to peers and others (NSTA 2004). Inquiry-based instruction is an effective early childhood science teaching strategy (Dejonckheere et al., 2016; McLean, Jones & Schaper, 2015; NRC, 2000; Trundle, 2009; Trundle & Sackes, 2015; Van Uum, Verhoeff & Peeters, 2017). Inquiry-based instructional approaches offer the most effective way for young children to engage with and learn science concepts. Children are expected to be active agents in learning activities, as they are engaged in activities such as work in small groups, meaningful science activities, and sense-making.

The findings in this study indicated that only eight participants (8%) conceptualized science instruction as being inquiry-based, that is, as involving science activities having both hands-on components and minds-on (cognitive activities) components. Among these eight

participants, the conception of science instruction as inquiry-based was associated with a specific science content and this was connected to a minds-on activity, such as "asking questions." The number of participants who conceptualized science instruction as inquiry-based was very low. Therefore, findings from this study indicate that participants in this study had limited prior knowledge of inquiry-based science instruction as a developmentally appropriate instructional strategy for children in pre-K to 3rd grade science classroom, an approach advocated by the literature (Dejonckheere et al., 2016; McLean, Jones & Schaper, 2015; NRC, 2000; NSTA 2004; Trundle, 2009; Trundle & Sackes, 2015; Van Uum, Verhoeff & Peeters, 2017). Furthermore, the low incidence of inquiry highlights a serious concern that must be addressed in early childhood methods courses, in all teacher education courses, and in content courses taken prior to enrollment in early childhood teacher education programs. For participants in this study, taking four science content courses (Biology for Educators, Environmental Science, Earth Science, and Conceptual Physics) provided only limited knowledge of inquiry-based science instruction. This further highlights the importance of engaging science content course instructors and early childhood teacher educators in collaboration on cross-curricular initiatives that will integrate inquiry-based strategies and enable prospective early childhood teachers, like the participants in this study, to experience inquiry-based instruction themselves.

Findings in Relation to the Sociocultural Theory

According to the sociocultural theory, children learn science through mediation within the zone of proximal development. The zone of proximal development is created by the teacher and student in the context of specific tasks for intersubjective agreement about meaning being made together: it is not simply announced by the teacher but rather is carried out through mutual participation (John-Steiner & Mahn, 1996; Subramaniam, 2007). This theory captures the idea of teachers as experts in the use of cultural tools leading to cultural amplification and the extending of students' cognitive processes. This theory also incorporates the idea of psychological tools (the semiotic tools of speech, writing, language, and thought) and physical tools as mediators of thought and activity within the zone of proximal development. Another essential element of the social interaction is scaffolding. In Vygotsky's view of cognitive development, the adults or other partners in a child's world should provide scaffolding to help children learn new information and develop more complex thinking abilities. In the classroom, the activity structure manifests as the "interactive opportunities" that enable the teacher to select the relevant and suggested content, organize the content and relate it to what their audience already knows. As a result, the teachers make it possible for themselves and their students to " publicly display thinking and reasoning and for their students to gain access to learning and to demonstrate social and academic competence" (Subramaniam, 2007; Weade, 1987: 17).

Findings from this study indicate that participants' conceptions play a role in the mediation process (John-Steiner & Mahn, 1996; Subramaniam, 2007), but the nature of the conceptions, that is, the dimensions which they subscribe to, social, implementation, cognitive, and affective, determine the mediation in the learning of science. The conceptions of science instruction that fell into the categorization of having passive and affective dimensions provided a glimpse into how participants intended to use "fun," "excitement," "motivation," etc., to mediate their future students' learning science content through demonstrations.

to enable their students to construct science content after mediation while following an affective approach. The findings indicated that the demonstrations with affective dimensions promoted passive learning through showing, telling, describing, etc. science content to students in a traditional teaching format without any student interaction and/or involvement with the demonstrations.

On the other hand, the conceptions of science instruction that fell into the categories associated with active learning and with social, implementation, and cognitive dimensions provided a glimpse into how participants intend to use mediation and how this mediation extends to students' learning of science content. The nature of mediation as seen through demonstrations, engagements, inquiry, observations, and doing experiments had built-in dimensions of social, implementation, and cognitive dimensions that called for students to be involved in an active process of learning the science content through mediation.

Unique Findings

Unique to this study was the finding that participants' conceptions of science instruction, using the dimensions from the STLP3 and the frameworks guiding the study, were complex in nature and multidimensional. The use of the STLP3 instrument revealed four dimensions underscoring participants' conceptions of science instruction: the *Affective* dimension, the *Cognitive* dimension, the *Implementation* dimension, and the *Social* dimension. This uniqueness made possible the further classification of participants' conceptions into categories of active learning and passive learning. Furthermore, the complexity, the multidimensionality, and the categorization arising from the results of this study provide a framework in which to situate participants' conceptions of science instruction. The complexity

of the framework is inherent in the situating of dimensions within the conceptions of science instruction and thus providing a lens through which to view teachers ' specifications, as well as propositions related to science teaching and learning. The multidimensionality of the framework is inherent in situating conceptions within dimensions and thus provides clarity on teachers' thinking about science teaching and learning. The categories (active, passive, and dimensions) in the framework identify the orientations inherent in teachers ' thinking about science teaching, and their strategies for pre-K to 3rd children's learning of science content.

Limitations

One limitation of this study is the issue of participants' artistic aptitude (Subramaniam, 2013; 2018). It must be noted that not all participants in this study may have had the ability to draw or depict certain images in their drawings of themselves teaching science in the early childhood classroom. This limitation was mitigated through the use of narratives. Participants were asked to write a five to ten sentence narrative about their drawings to clarify what the products and processes depicted in the drawings were (Subramaniam, 2013; 2018). Another limitation of this study involves, the issue of power relations between the participants of this study and the researcher who collected the data in the form of drawing and narratives, who was, an authority figure. The issue of power relations was mitigated because the researcher who collected the data in the early childhood teacher education program, and he collected the data in the presence of the early childhood methods instructors.

Educational Implications

Implications for the Early Childhood Teacher Educator

The findings of the study provide a framework for focusing on how prospective early

childhood teachers conceptualize science instruction for children in pre-K to 3rd grade classrooms and their strategies for helping their future early childhood learners in constructing science content. Early childhood teacher educators must realize that the teaching process is becoming an important variable in the development of their prospective early childhood teachers' epistemological belief, and is now a crucial factor for opening doors to new vistas and stimulating novel alternative solutions and ideas in them. Teacher educators may benefit by helping prospective early childhood teachers to explore their teaching and learning conceptions regarding the students in their classes (Mahasneh, 2018) and to use the resulting knowledge to develop lesson planning and more effective pedagogy in the early childhood science methods courses.

Implications for the Early Childhood Teacher Preparation Programs

The curriculum for early childhood teacher education must incorporate frameworks for understanding the practice of teaching within coursework early on in the early childhood teacher preparation program to enable prospective teachers to reflect on and document their prior knowledge of science instruction. By doing so, it will enable prospective early childhood teachers to look into their own and their peers' complex and multidimensional conceptions of science instruction and situate their prior and new knowledge of early childhood science instruction within theoretical frames rather than simply relying on their knowledge of science instruction from K-12 experiences.

Implications for Future Research

The findings of this study call for more research into the prior knowledge of prospective

early childhood teachers. Specifically, future research should study conceptions of science instruction in relation to conceptions of teachers prior to and after the end of early childhood methods courses. From a personal perspective, I would like to compare the conceptions of science instruction between different demographics of early childhood educators, for example between early childhood educators from Thailand and from other countries, such as the U.S.A. Additionally, there is a need for studies that examine early childhood educators' conceptions of science instruction from a multimodal (interviews, metaphors, focus groups, etc.) perspective. Through such work, a holistic understanding of this phenomenon will become possible.

Conclusion

Preparing prospective early childhood teachers to teach science effectively in early childhood classrooms is becoming more important, particularly since science methods courses taken by prospective early childhood teachers will influence their future instruction in early childhood classrooms (Harrell & Subramaniam, 2015; Subramaniam, 2018). In summary, I find that prospective early childhood teachers entering a teacher educating program possess a variety of conceptions about science, teaching, and learning. When they enroll in early childhood education courses, it is important that these conceptions be made explicit, that they be analyzed and discussed critically, and that beginning early childhood teachers be provided with other theoretically-based conceptions of science, teaching, and learning for children in pre-K to 3rd grade classrooms.

APPENDIX A

IRB



THE OFFICE OF RESEARCH AND INNOVATION Research and Economic Development

February 26, 2019

PI: Karthigeyan Subramaniam Study Title: Prospective Early Childhood Teachers' Conceptions of Science Instruction

RE: Human Subjects Application # IRB-19-51

Dear Dr. Karthigeyan Subramaniam:

In accordance with 45 CFR Part 46 Section 46.104, your study titled "Prospective Early Childhood Teachers' Conceptions of Science Instruction" has been determined to qualify for an exemption from further review by the UNT Institutional Review Board (IRB).

No changes may be made to your study's procedures or forms without prior written approval from the UNT IRB. Please contact The Office of Research Integrity and Compliance at 940-565-4643 if you wish to make any such changes. Any changes to your procedures or forms after 3 years will require completion of a new IRB application.

We wish you success with your study.

Sincerely,

Allelley have

Shelley Riggs, Ph.D. Professor Chair, Institutional Review Board

SR:jm

APPENDIX B

SCIENCE TEACHING AND LEARNING PORTRAYALS OF PROFESSIONAL PRACTICE (STLP3)

SCORING RUBRIC

Science Teaching and Learning Portrayals of Professional Practices (STLP3) Scoring Rubric Experience excitement, interest and motivation to learn about phenomena in the natural and physical world [affective] (Goal 1)

Score Description

- 4 Smiling figures with specific indicators such as the use of descriptive words "fun" or other exclamations of excitement, interest and motivation
- 3 Smiling figures, but no specific indicators of excitement, interest and motivation
- 2 Figures with facial expressions but ambiguous in regard to excitement, motivation, and interest
- 1 Negative facial expressions or comments suggesting lack of interest or motivation
- 0 No evidence (facial expression or comments) of excitement, interest, or motivation in the drawing

Come to generate, understand, remember and use concepts, explanations, arguments, models and facts related to science [cognitive] (Goal 2)

Score Description

- 4 Evidence in thought bubbles, comments, or models of concepts, explanations, arguments, models, or facts (4 or more present)
- 3 Evidence in thought bubbles, comments, or models of concepts, explanations, arguments, models, or facts (3 present)
- 2 Evidence in thought bubbles, comments, or models of concepts, explanations, arguments, models, or facts (2 present)
- 1 Evidence in thought bubbles, comments, or models of concepts, explanations, arguments, models, or facts (1 present)
- 0 No evidence of concepts, explanations, arguments, models, or facts present

Manipulate, test, explore, predict, question, observe and make sense of the natural and physical world [implementation] (Goal 3)

Score Description

4 Evidence in thought bubbles, comments, or activities of manipulating, testing, exploring, predicting, questioning, observing, or sense-making (4 or more present)

- 3 Evidence in thought bubbles, comments, or activities of manipulating, testing, exploring, predicting, questioning, observing, or sense-making (3 present)
- 2 Evidence in thought bubbles, comments, or activities of manipulating, testing, exploring, predicting, questioning, observing, or sense-making (2 present)
- 1 Evidence in thought bubbles, comments, or activities of manipulating, testing, exploring, predicting, questioning, observing, or sense-making (1 present)
- 0 No evidence of manipulating, testing, exploring, predicting, questioning, observing, or sense-making

Participate in scientific activities and learning practices with others, using scientific language and tools [social] (Goal 5)

Score Description

- 4 Evidence of learning with others, using scientific language and using scientific tools. 2 or more present)
- 3 Evidence of learning with others, using scientific language and using scientific tools. (1 present)
- 2 Ambiguous connection to science
- 1 Students not participating in a science activity or practice
- 0 No evidence of learning with others, using scientific language, or using scientific tools

APPENDIX C

SCORING RUBRIC SUPPLEMENTAL INFORMATION SHEET

Scoring Rubric Supplemental Information Sheet

Goal 1: Experience excitement, interest and motivation to learn about phenomena in the natural and physical world.

-Look to the mouth on the figures present. If anyone is smiling, give credit (that is, if only the teacher is smiling but the students are not, give credit for smiling or the reverse). If faces are not visible, look for specific indicators of excitement, interest and motivation in thought bubbles or comments.

Excitement - thought bubbles or comments expressing excitement (e.g., exclamation marks)

Interest - thought bubbles or comments about what is occurring

Motivation - thought bubbles or comments expressing eagerness (e.g., "I can't wait to do this", "Let's get started")

Goal 2: Come to generate, understand, remember and use concepts, explanations, arguments, models and facts related to science.

-Identify concepts, explanations, arguments, models and facts using these descriptions.

Concepts - thought bubbles or comments about bigger science ideas (e.g. energy, evolution)

Explanations - thought bubbles or comments about how things are happening

Arguments - thought bubbles or comments that compare or respond to alternatives

Models - a visual model of three dimensions related to scientific phenomena (not classroom management)

Facts - a statement of science learning (e.g. deciduous trees lose leaves in the fall here)

Goal 3: Manipulate, test, explore, predict, question, observe and make sense of the natural and physical world.

-Identify manipulating, testing, exploring, predicting, questioning, observing, and sense-making using these descriptions:

Manipulating - each learner has access to materials in reach or is shown actually touching items (note: manipulating variables for an experiment, see testing below)

Testing - thought bubbles or comments that illustrate a trial ("what will happen if..."); presence of testing tools (manipulating variables for an experiment)

Exploring - engaged in active science (not only reading books and writing)

Predicting - thought bubbles or comments stating what might happen

Questioning - students have question marks or actual questions visible

Observing - looking intently as individuals or groups at an object or phenomena

Sense-making-thought bubbles or comments that indicate students or the teacher are trying to "figure things out," phrases that begin with "maybe"...

Goal 5: Participate in scientific activities and learning practices with others, using scientific language and tools.

-Identify participating in scientific activities and learning practices with others, using scientific language, and tools using these descriptions:

Participate in scientific activities and learning practices with others - students grouped for interaction

Scientific language - use of terms associated with science (such as, comparisons, questions about how)

Tools - clearly drawn or unclearly drawn (squiggles) materials available to all learners

http://www.drawntoscience.org/researchers/scoring-rubric/rubric-instructions.html

APPENDIX D

EXAMPLES OF PARTICIPANTS' DRAWINGS

CD 600 181 my drawing is showing a teacher that is using a visual poster to help describe where photo synthesis is. She has brought in different plants & flowers to show the kids what can happen to the plants & explaining how they ear She also has facts on the white board & another diagram as well. She also has a little book with plams that "eat" pictures it enough touch those who don't &



- children ave learning about the life cycle of a plant & how it utilizer -The teacher unite unite hovers and adding dy, to the soil to demonstrate how water is abioined through NO DIE Thus the Red Py Yellos Prthe root. She wears SODDODAN safety - She engages the children by allowing them to water the Please turn over the prints. Please turn over the page if additional space is needed. She speaks through out the whole proved explaining everyming.







The science roome has a demonstration Voicanos!! after a full day of teachy WON about voleares to (that) Sive kids a visual (100) of how ulcanos WORK . She follows all suffer cani TORO guidline & her LAVA) tion stay in seats. She for them Stay behind the sarry lins I wear screety gosser. Wana tory She has themeter of when's lunch glores & leeb coat to show safetes. All pide run it but Some lost but are intrigued.



REFERENCES

- Aguilar, N. A. (2016). Examining the integration of science, technology, engineering, and mathematics (STEM) in preschool and transitional kindergarten (TK) classrooms using a social-constructivist approach (Order No. 10111641). Available from ProQuest Dissertations & Theses Global; SciTech Premium Collection; Social Science Premium Collection. (1799589223). Retrieved from https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/17995 89223?accountid=7113
- Akcay, N. O. (2016). Determining the views and adequacy of the Preschool Teachers Related to Science Activities. Universal Journal of Educational Research, 4(4), 821-829. DOI: 10.13189/ujer.2016.040419
- Akcay, N. O. (2016). Implementation of Cooperative Learning Model in Preschool. *Journal of Education and Learning, 5(*3), doi:10.5539/jel.v5n3p83 URL: http://dx.doi.org/10.5539/jel.v5n3p83
- Allen, B. (2017). Exploring the role of ideology in interdisciplinary science education policy, *Educational Studies*, *53*(6), 642-653. DOI: 10.1080/00131946.2017.1369081.
- Ambusaidi, A., & Al-Balushi, S. M. (2012) A longitudinal study to identify prospective science teachers' beliefs about science teaching using the draw-a-science-teacher-test checklist. *International Journal of Environmental and Science Education*, 7(2),291-311. <u>https://www.researchgate.net/publication/260025010</u>
- Andersson, K., & Gullberg, A., (2014). What is science in preschool and what do teachers have to know to empower children? *Cultural Studies of Science Education, 9*(2), 275-296. doi:10.1007/s11422-012-9439-6
- Antoniadou, P., & Skoumios, G. M. (2013). Primary teachers' conceptions about science teaching and learning. *International Journal of Science in Society*, 4(1), 69–82. Retrieved from <u>https://libproxy.library.unt.edu:9443/login?url=http://search.ebscohost.com/login.aspx</u> ?direct=true&db=a9h&AN=91821722&scope=site
- Areljung, S., Ottander, C., & Due, K. (2017). 'Drawing the leaves anyway': Teachers embracing children's different ways of knowing in preschool science practice. *Research in Science Education*, 47(6), 1173-1192. doi: http://dx.doi.org/10.1007/s11165-016-9557-3
- Arthur, J., Waring, M., Coe, R., & Hedges, L. V. (Eds.). *Research methods and methodologies in education*. Los Angeles: Sage.

- Ashbrook, P. (2018). Early childhood science and engineering resources. Retrieved from <u>https://www.earlychildhoodwebinars.com/wp-content/uploads/2016/05/Early-</u> Childhood-Science-and-Engineering-Resources-.pdf
- Assiri, Y. I. (2016). Science teachers' knowledge, beliefs, values, and concerns of teaching through inquiry (Order No. 10194135). Available from ProQuest Dissertations & Theses Global; Social Science Premium Collection. (1871028150). Retrieved from <u>https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/18710</u> 28150?accountid=7113
- Beltman, S., Glass, C., Dinham, J., Chalk, B. & Nguyen, B. (2015). Drawing identity: Beginning pre-service teachers' professional identities. *Issues in Educational Research*, 25(3), 225-245. <u>http://www.iier.org.au/iier25/beltman.html</u>
- Blanck, G. (1990). Vygotsky: The man and his cause. In L.C. Moll (Ed.), *Vygotsky and education* (pp. 31-58). Cambridge University Press.
- Blevins, K. (2018). *Kindergarten teachers' understanding of the elements of implementing inquiry-based science instruction* (Order No. 10743086). Available from ProQuest Dissertations & Theses Global. (2002070601). Retrieved from https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/2002070601?accountid=7113
- Boutte, G. S. (2018). Taking Culturally Relevant Teaching to the Big House: Implications for early childhood teacher educators. *New Educator, 14*(2), 171-184. doi:10.1080/1547688X.2018.1426327
- Bransford, J. D., Brown, A. L., & Cocking, R. R. (2000). *How people learn: Brain, mind, experience, and school* (Expanded ed.). Washington, D.C.: National Academies Press.
- Brenneman, K. (2011). Assessment for preschool science learning and learning environments. *Early Childhood Research and Practice*, 13(1). <u>http://ecrp.uiuc.edu/v13n1/brenneman.html</u>
- Brenneman, K., Stevenson-Boyd, J., & Frede, E. C. (2009). Mathematics and science in preschool: Policy and practice. *New Brunswick, NJ: National Institute for Early Education Research*.
- Brooks, M. (2009). Drawing, visualsation and young children's exploration of "Big Ideas". International Journal of Science Education, 31(3), 319-341. DOI: 10.1080/09500690802595771.
- Brown, C.P. & Englehardt, J. (2017). A case study of how a sample of preservice teachers made sense of incorporating iPads into their instruction with children. *Journal of Early Childhood Teacher Education, 38*(1), 19-38. DOI: 10.1080/10901027.2016.1274695

- Brown, M. H., & Schwartz, R. S. (2009). Connecting photosynthesis and cellular respiration: preservice teachers' conceptions. *Journal of Research in Science Teaching*, *46*(7), 791-812.
- Buldu, M. (2006). Young children's perceptions of scientists: A preliminary study. *Educational Research*, 48(1), 121-132. DOI: 10.1080/00131880500498602
- California Department of Education, 2012. In F. Ong (Ed.), *The California preschool learning foundations*, Vol. 3. Sacramento, CA: California Department of Education.
- Can, B., Yıldız-Demirtaş, V., & Altun, E. (2017). The effect of project based science education programme on scientific process skills and conceptions of kindergarten students. *Journal of Baltic Science Education*, *16*(3), 395-413.
- Carter, G., Westbrook, S. L. & Thompkins, C. D. (1999). Examining science tools as mediators of students' learning about circuits. *Journal of Research in Science Teaching, 36,* 1, 89–106.
- Chan, K.W. & Elliot, R.G. (2004). Relational analysis of personal epistemology and conceptions about teaching and learning. *Teaching and Teacher Education, 20*, 817–831
- Chang, N. (2012). The role of drawing in young children's construction of science concepts. *Early Childhood Education Journal, 40*(3), 187-193. doi: http://dx.doi.org/10.1007/s10643-012-0511-3
- Çil, E. (2014). Teaching nature of science to pre-service early childhood teachers through an explicit reflective approach. Asia - Pacific Forum on Science Learning and Teaching, 15(1), 1-21. Retrieved from <u>https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/19559</u> 03373?accountid=7113
- Clements, D., & Sarama, J. (2016). Math, science, and technology in the early grades. *The Future* of Children, 26(2), 75-94. Retrieved from <u>http://www.jstor.org/stable/43940582</u>
- Coffey, A., & Atkinson, P. (1996). *Making sense of qualitative data: Complementary research strategies.* Thousand Oaks, CA, US: Sage Publications, Inc.
- Coffin, T. B. (2012). Science education in the early childhood & special education setting: An analysis of science education across lake Washington, Bellevue & Seattle school districts (Order No. 1515880). Available from ProQuest Dissertations & Theses Global. (1034575537). Retrieved from <u>https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/10345</u> 75537?accountid=7113
- Conezio, K. & French, L, (2002). Science in the preschool classroom. *Young Children, 0,* 1-5. Retrieved from file:///F:/SCIENCE%20in%20the%20preschool%20classroom.pdf
- Confrey, J. (1995). How compatible are radical constructivism, sociocultural approaches, and social constructiotivism? In L.P. Steffe & J. Gale (Eds.), *Constructivism in education* (pp. 185-225). New Jersey: Lawrence Erlbaum Associates, Publishers.
- Connelly, F.M., & Clandimin, D.I. (1986). On narrative method, personal philosophy, and narrative unities is the story of teaching. *Journal of Research in Science Teaching, 23*(4), 293-310.
- Copple, C., & Bredekamp, S. (2009). Developmentally appropriate practice in early childhood programs: Serving Children from Birth through Age 8 (3rd ed.). Washington, DC: NAEYC.
- Creswell, J. W. (2009). Research design: Qualitative, quantitative, and mixed methods approaches (3rd Ed.). Sage Publications. Thousand Oaks, CA 91320.
- Curtis, D. (2002). The power of projects. *Educational Leadership*, 60(1), 50-54.

Dejonckheere, P. J. N., De Wit, N., Van, d. K., & Vervaet, S. (2016). Exploring the classroom: Teaching science in early childhood. *International Electronic Journal of Elementary Education, 8*(4), 537-557. Retrieved from https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/18077 40291?accountid=7113

Delserieys, A., Impedovo, M., Fragkiadaki, G., & Kampeza, M. (2017). Using drawings to explore children's preschool's ideas about shadow training. *Review of Science Mathematics & ICT Education, 11(*1), 55-69. Retrieved from file:///D:/Exiting%20Researches%20on%20Early%20Childhood%20Science%20Education /Delserieys,%20A.,%20Impedovo,%20M.,%20Fragkiadaki,%20G.,%20&%20Kampeza,%2 0M.%20(2017).%20Using%20drawings.pdf

- Eckhoft, A. (2017). Partners in Inquiry: A collaborative life science investigation with preservice teachers and kindergarten students. *Early Childhood Education Journal, 45,219*–227. DOI 10.1007/s10643-015-0769-3
- Edwards, S. (2006). "Stop thinking of culture as geography": early childhood educators' conceptions of sociocultural theory as an informant to curriculum. *Contemporary Issues in Early Childhood*, 7(3), 238–252. Retrieved from <u>https://libproxy.library.unt.edu:9443/login?url=http://search.ebscohost.com/login.aspx</u> ?direct=true&db=a9h&AN=22953589&scope=site

Edwards, K., & Loveridge, J. (2011). The inside story: Looking into early childhood teachers' support of children's scientific learning. *Australian Journal of Early Childhood, 36*(2), 28-35. Retrieved from <u>file:///E:/Constructivist/Edwards,%20K.,%20&%20Loveridge,%20J.%20(2011).%20%20T</u> <u>he%20inside%20story%20Looking%20into%20early%20childhood%20teachers'%20supp</u> <u>ort%20of%20children's%20scientific%20learning.pdf</u>

- Elmas, D., Demirdogen, B., & Geban, o. 2011. "Preservice chemistry teachers' Images about science teaching in their future classrooms." *H. U. Journal of Education, 40*, 164–175.
- Elo, S. & KYNGA[°]S, H. (2008). The qualitative content analysis process. *Journal of Advanced Nursing*, *62*(1), 107–115doi: 10.1111/j.1365-2648.2007.04569.
- Elo, S., Kääriäinen, M., Kanste, O., Pölkki, T., Utriainen, K., & Kyngäs, H. (2014). Qualitative content analysis: a focus on trustworthiness. SAGE Open. <u>https://doi.org/10.1177/2158244014522633</u>
- Escamilla, I. i., Khales, B. b., Meier, D. d., & Melgoza, M. (2018). Reflecting across borders: Palestinian and US early childhood educators engage in collaborative science inquiry. *YC: Young Children*, *73*(2), 67-73.
- Finson, K.D. (2001). Investigating preservice elementary teachers' self-efficacy relative to selfimage as a science teacher. *Journal of Elementary Science Education*, 13(1), 31-41. https://doi.org/10.1007/BF03176931
- Fitzgerald, A., & Schneider, K. (2013). What teachers want: Supporting primary school teachers in teaching science. *Teaching Science: The Journal of the Australian Science Teachers Association*, 59(2), 7–10. Retrieved from <u>https://libproxy.library.unt.edu:9443/login?url=http://search.ebscohost.com/login.aspx</u> <u>?direct=true&db=a9h&AN=89641551&scope=site</u>
- Fleer, M., & Pramling, N. (2014). A cultural-historical study of children learning science: foregrounding affective imagination in play-based settings. Springer Netherlands. ProQuest Ebook Central, http://ebookcentral.proquest.com/lib/unt/detail.action?docID=1968262
- Fleer, M. & Robbins, J. (2003). "Hit and run research" with "hit and miss" results in early childhood science education. *Research in Science Education*, 33(4), 405-431. Retrieved from <u>file:///E:/Dr.%20Castro/Hit%20and%20Run%20Research.pdf</u>
- Frawley, W. (1997). *Vygotsky and cognitive science*. Cambridge: Harvard University Press.
- French, L. (2004). Science as the center of a coherent, integrated early childhood curriculum. *Early Childhood Research Quarterly, 19*, 138-149.
- Fusaro, M. & Smith, M. C. (2018). Preschoolers' inquisitiveness and science-relevant problem solving. *Early Childhood Research Quarterly*, 42, 119-127. <u>https://doi.org/10.1016/j.ecresq.2017.09.002</u>
- Garbett, D. (2003). Science Education in Early Childhood Teacher Education: Putting Forward a Case to Enhance Student Teachers' Confidence and Competence. *Research in Science Education, 33,* 467-481. https://doi.org/10.1023/B:RISE.0000005251.20085.62

- Gelman, R., & Brenneman, K. (2004). Science learning pathways for young children. *Early Childhood Research Quarterly*, *19* (1), 1st Quarter, 150-158. https://doi.org/10.1016/j.ecresq.2004.01.009
- Gerde, H. K., Schachter, R. E., & Wasik, B. A. (2013). Using the scientific method to guide learning: An integrated approach to early childhood curriculum. *Early Childhood Education Journal*, 41(5), 315-323. doi: <u>http://dx.doi.org/10.1007/s10643-013-0579-4</u>
- Gilbert, A. (2009). Utilizing Science Philosophy Statements to Facilitate K-3 Teacher Candidates' Development of Inquiry-based Science Practice. *Early Childhood Education Journal*, 36(5), 431-438. doi:10.1007/s10643-009-0302-7
- Gillmeister, K. M. (2017). Development of early conceptions in systems thinking in an environmental context: An exploratory study of preschool students' understanding of stocks & flows, behavior over time and feedback (Order No. 10256359). Available from ProQuest Dissertations & Theses Global. (1925350861). Retrieved from <u>https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/19253</u> <u>50861?accountid=7113</u>
- Go, Y., & Kang, J. (2015). Early childhood pre-service teachers' self-images of science teaching in constructivism science education courses. *Asia Pacific Forum on Science Learning and Teaching*, *16*(2), 1-25. Retrieved from <a href="https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu/login?url=https://libproxy.library.unt.edu?165/docview
- Goldstein, L. (2008). Teaching the Standards is Developmentally Appropriate Practice: Strategies for Incorporating the Sociopolitical Dimension of DAP in Early Childhood Teaching. *Early Childhood Education Journal*, *36*(3), 253–260. <u>https://doi.org/10.1007/s10643-008-0268-x</u>
- Graue, M. E., Whyte, K. L. & Karabon, A. E. (2015). The power of improvisational teaching. *Teaching and Teacher Education, 48,* 13-21. Retrieved from https://ac.els-cdn.com/S0742051X15000153/1-s2.0-S0742051X15000153-main.pdf? tid=f744ff6a-2a31-42aa-8c56d6eb5f02fa46&acdnat=1521759793 203fb9e62ac39c7f81eebd3b59bf8646
- Grbich, Carol (2013). *Qualitative data analysis: An introduction (2nd ed*.). Thousand Oaks, CA: SAGE
- Greenfield, D.B., Jirout, J., Dominguez, X., Ariela Greenberg, A., Maier, M., & Fuccillo, J. (2009).
 Science in the Preschool Classroom: A Programmatic Research Agenda to Improve
 Science Readiness. *Early Education and Development, 20*:2, 238-264. DOI: 10.1080/10409280802595441

- Grigg, W., Lauko, M., & Brockway, D. (2006). *The Nationis Report Card: Science 2005 (NCES 2006-466)*. U.S. Department of Education, National Center for Education Statistics. Washington, D.C.: U.S. Government Printing Office.
 https://nces.ed.gov/nationsreportcard/pdf/main2005/2006466.pdf
- Guo, Y., Wang, S., Hall, A., Breit-Smith, A., & Busch, J. (2016). The Effects of Science Instruction on Young Children's Vocabulary Learning: A Research Synthesis. *Early Childhood Education Journal*, 44(4), 359-367. doi:10.1007/s10643-015-0721-6
- Harrell, P. & Subramaniam, K. (2015). Elementary pre-service teachers' conceptual understanding of dissolving: A Vygotskian concept development perspective. *Research in Science and Technological Education*, *33*(3), 304-324.
- Haney, W., Russell, M., & Bebell, D. (2004). Drawing on Education: Using Drawings to Document Schooling and Support Change. *Harvard Educational Review*, 74(3),241-272.<u>https://doi.org/10.17763/haer.74.3.w0817u84w7452011</u>
- Hassard, J. & Dias, M. (2009). *The Art of Teaching Science: Inquiry and Innovation in Middle School and High School* (2nd ed.). New York, Routledge.
- Hennig, K., & Kirova, A. (2012). The Role of Cultural Artefacts in Play as Tools to Mediate Learning in an Intercultural Preschool Programme. *Contemporary Issues in Early Childhood*, 13(3), 226–241. <u>https://doi.org/10.2304/ciec.2012.13.3.226</u>

Hesse-Biber, S.N. (2017). The Practice of Qualitative Research (3rd ed.). Thousand Oaks, CA:

- Hodson, D., & Hodson, J. (1998a). From constructivism to social constructivism: A Vygotskian perspective on teaching and learning science. *School Science Review*, *79*(289), 33-41.
- Hodson, D., & Hodson, J. (1998b). Science Education as enculturation: Some implication for practice. *School Science Review*, *80*(290), 17-23.
- Hoffman, J. L., Collins, M. F., & Schickedanz, J. A. (2015). Instructional Challenges in Developing Young Children's Science Concepts. *Reading Teacher*, 68(5), 363-372. doi:10.1002/trtr.1325
- Hoisington, C., & Winokur, J. (2015). Tools of Science Inquiry That Support Life Science Investigations. Science & Children, 52(6), 24-25. Retrieved from <u>file:///D:/Science%20Images/Hoisington,%20C.,%20&%20Winokur,%20J.%20(2015).%20</u> <u>%20Tools%20of%20Science%20Inquiry.pdf</u>
- Hollweg, K.S., &Hill, D. (2003). What Is the Influence of the National Science Education Standards? Reviewing the Evidence, A Workshop Summary. Washington, D.C.: National Academies Press.

- Hong, S.Y., Torquati, J., & Molfese, V. J. (2013). "Theory Guided Professional Development in Early Childhood Science Education." CYFS. Faculty *Publications*. Retrieved from http://digitalcommons.unl.edu/cyfsfacpub
- Howe, A. C. (1996). Development of science concepts within a Vygotskian framework. *Science Education*, *80*(1), 35-51.
- Howes, E.V. (2008). Educative experiences and early childhood science education: A Deweyan perspective on learning to observe. *Teaching and Teacher Education, 24*, 536-549. doi:10.1016/j.tate.2007.03.006.
- Howitt, C. (2011). Planting the Seeds of Science. Development and evaluation of a new flexible and adaptable early childhood science resource. *Teaching Science: The Journal of the Australian Science Teachers Association*, 57(3), 32-39.
- Hsieh, W., & Tsai, C. (2017). Exploring students' conceptions of science learning via drawing: a cross-sectional analysis. *International Journal of Science Education*, 39(3), 274-298, DOI: 10.1080/09500693.2017.1280640 <u>https://doi.org/10.1080/09500693.2017.1280640</u>
- Inan, H.Z. & Inan, T. (2015). 3Hs Education: Examining hands-on, heads-on and hearts-on early childhood science education. *International Journal of Science Education*, *37*(12), 1974-1991, DOI: 10.1080/09500693.2015.1060369
- Institute for the Promotion of Teaching Science and Technology, The (IPST), (2011). *Science Standards Framework for Early Childhood Education*. Bangkok Khurusapa Publishing House, Ladprao. Retrieved from <u>http://www.ipst.ac.th</u>
- Jaipal-Jamani, K., & Angeli, C. (2017). Effect of Robotics on Elementary Preservice Teachers' Self-Efficacy, Science Learning, and Computational Thinking. *Journal of Science Education & Technology*, *26*(2), 175-192. doi:10.1007/s10956-016-9663-z
- Jinan, k. b. (2016). Children Draw the World to Know the World. *Exchange (19460406)*, (229), 61–64. Retrieved from https://libproxy.library.unt.edu:9443/login?url=http://search.ebscohost.com/login.aspx ?direct=true&db=eue&AN=115879319&scope=site
- John-Steiner, V., & Mahn, H. (1996). Sociocultural approaches to learning and development: A Vygotskian framework. *Educational Psychologist*, *31*(3/4), 191-206.
- Jones, I., Lake, V. E., & Dagli, U. (2005). Integration of science and mathematics methods and preservice teachers' understanding of constructivism. *Journal of Early Childhood Teacher Education*, 25(2), 165-172. doi:10.1080/1090102050250209
- Katz, L. (2018). All about Balls: A Preschool Project. Retrieved from <u>https://www.researchgate.net/publication/237267336</u> All about Balls A Preschool P <u>roject</u>

- Katz, L. G., & Chard, S. C. (2000). *Engaging children's minds: The project approach* (2nd ed.). Stamford, CT: JAI Press.
- Katz, P., McGinnis, R. J., Hesrness, E., Riedinger, K., Marback-Ad, G., Dei, A., & Pease. R, (2011). Professional identity development of teachers' candidates participating in on informal science education internship: A focus on drawings as evidence. *International Journal of Science Education*, 33(9), 1169-1197. doi: .10, 1080/09500693,20 10.489928
- Kim, H. K., & Han, H. S. (2015). Understanding early childhood teachers' beliefs and self-stated practices about social competence instructional strategies in the context of developmentally appropriate practice: a comparison of preservice and in-service teachers in the United States. *European Early Childhood Education Research Journal*, 23(4), 476-496. doi:10.1080/1350293X.2015.1087152
- Kirschner, P., Sweller, J., & Clark, R. (2006). Why minimal guidance during instruction does not work: An analysis of the failure of constructivist, discovery, problem-based, experimental and inquiry-based teaching. *Educational Psychologist, 40*, 75-86.
- Koren, P. & Bar, V. (2009). Pupils' Image of 'the Scientist' among Two Communities in Israel: A comparative study. *International Journal of Science Education*, 31(18), 2485-2509. DOI: 10.1080/09500690802449375
- Kostelnik, M.J., Soderman, A.K., & Whiren, A.P. (2011). *Developmentally Appropriate Curriculum: Best Practices in Early Childhood Education* (5th ed.). Pearson Education, Boston, M.A.
- Kozulin, A. (1990). *Vygotsky's psychology: A biography of ideas*. Cambridge: Harvard University Press.
- Kurasaki, K.S. (2000). Intercoder reliability for validating conclusions drawn from open-ended interview data. *Field Methods*, *12*(3), 179-194. Doi:10.1177/1525822X0001200301
- Kuru, N., & Akman, B. (2017). Examining the science process skills of preschoolers with regards to teachers' and children' variables. *Egitim Ve Bilim, 42*(190) Retrieved from <u>https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/19004</u> <u>88715?accountid=7113</u>
- Lee, L., & Tu, X. (2016). Digital Media for Low-Income Preschoolers' Effective Science Learning: A Study of iPad Instructions with a Social Development Approach. *Computers in the Schools, 33*(4), 239–252. https://doi.org/10.1080/07380569.2016.1249742
- Lee, J. & Yoon, J. Y. (2008). Teaching Early Childhood Teacher Candidates How to Assess Children's Inquiry Skills in Science Learning. *Contemporary Issues in Early Childhood,* 9(3), 265-269. Retrieved from http://journals.sagepub.com/doi/pdf/10.2304/ciec.2008.9.3.265

- Lemke, J. L. (2001), Articulating communities: Sociocultural perspectives on science education. J. Res. Sci. Teach., 38: 296–316. doi:10.1002/1098-2736(200103)38:3<296::AID-TEA1007>3.0.CO;2-R.
- Mahasneh, A. M. (2018). Investigation of the Relationship between Teaching and Learning Conceptions and Epistemological Beliefs among Student Teachers from Hashemite University in Jordan. *European Journal of Contemporary Education*, 7(3), 531–540. <u>https://doi.org/10.13187/ejced.2018.3.531</u>
- Maier, M.F., Greenfield, D.B., & Bulotsky-Shearer, R. J. (2013). Development and validation of a preschool teachers' attitudes and beliefs toward science teaching questionnaire. *Early Childhood Research Quarterly, 28*, 366-378. https://doi.org/10.1016/j.ecresq.2012.09.003
- Markic, S., & Eilks. L. (2013). Potential changes in prospective chemistry teachers' beliefs about teaching and learning a cross-level study. *International Journal of Science and Mathematics Education*, *11*(4), 979-998. doi: 10,10071 s10763-013-9417-9
- Martin, D., Jean-Sigur, R., & Schmidt, E. (2005). Process-Oriented Inquiry—A Constructivist Approach to Early Childhood Science Education: Teaching Teachers to Do Science. *Journal of Elementary Science Education*, *17*(2), 13-26. Retrieved from http://www.jstor.org/stable/43156149
- Martina S. J. van Uum, Roald P. Verhoeff & Marieke Peeters (2017) Inquiry based science education: scaffolding pupils' self-directed learning in open inquiry. *International Journal of Science Education*, 39(18), 2461-2481, DOI: 10.1080/09500693.2017.1388940
- Mayring, P. (2000). Qualitative content analysis. *Forum: Qualitative Social Research, 1*(2). Retrieved from http://217.160.35.246/fqs-texte/2-00/2-00mayring-e.pdf
- McClure, E. R., Guernsey, L., Clements, D. H., Bales, S. N., Nichols, J., Kendall-Taylor, N., & Levine, M. H. (2017). STEM Starts Early: Grounding Science, Technology, Engineering, and Math Education in Early Childhood. In *Joan Ganz Cooney Center at Sesame Workshop.* Joan Ganz Cooney Center at Sesame Workshop. 1900 Broadway, New York, NY 10023.
- McKinnon, M., & Perara, S. (2015). The big picture: Pre-service teachers' perceptions of 'expert' science teachers. *Teaching Science: The Journal of the Australian Science Teachers Association*, 61(4), 32-45. Retrieved from file:///D:/Science%20ec%20preservice/McKinnon,%20M.,%20&%20Perara,%20S.%20(20 15).%20The%20big%20picture.%20Teaching%20Science%20The%20Journal%20Of%20T he%20Australian.pdf
- McLean, K., Jones, M., & Schaper, C. (2015). Children's literature as an invitation to science inquiry in early childhood education. *Australasian Journal of Early Childhood, 40*(4), 49-

56. Retrieved from

http://content.ebscohost.com/ContentServer.asp?T=P&P=AN&K=111938052&S=R&D=e ue&EbscoContent=dGJyMNHX8kSep684zOX0OLCmr0%2Bep65Ssae4SrKWxWXS&Conte ntCustomer=dGJyMPGqsE60rrVKuePfgeyx44Dt6fIA

- Medlin, D. (2017). Integrating Comparative Research on Global Instructional Practices in Pre-Service Early Childhood Education Science Course Instruction. International Research & Review: Journal of the Phi Beta Delta Honor Society for International Scholars, 6(2), 64-84. Retrieved from Medlin, %20D.%20(2017).%20Integrating%20Comparative%20Research%20on%20Global%20Ins tructional%20Practices%20in%20Pre-Service%20Early%20Childhood%20Education%20Science%20Course%20Instruction..pdf
- Miller, P. H. (1996). Theories of Developmental Psychology (3rd ed). New York: W.H. Freeman and Company.
- Ministry of Education. (2003). *Early Childhood Curriculum B.E. 2546 for Children of 3-5 Years Old.* Bangkok: Khurusapa Ladprao
- Minogue, J. (2010). What is the teacher doing? what are the students doing? an application of the draw-a-science-teacher-test. *Journal of Science Teacher Education, 21*(7), 767-781. doi: http://dx.doi.org/10.1007/s10972-009-9170-7
- Milford, T. M., & Tippett, C. D. (2013). Preservice teachers' images of scientists: Do prior science experiences make a difference? *Journal of Science Teacher Education, 24*(4), 745-762. doi: http://dx.doi.org/10.1007/s10972-012-9304-1
- National Association for the Education of Young Children (NAEYC). (2009). *Developmentally Appropriate Practice in Early Childhood Programs Serving Children from Birth through Age 8.* Retrieved from <u>https://www.naeyc.org/sites/default/files/globally-</u> <u>shared/downloads/PDFs/resources/position-statements/PSDAP.pdf</u>
- National Association for the Education of Young Children (NAEYC). (2012). 2010 NAEYC Standards for Initial & Advanced Early Childhood Professional Preparation Programs. Retrieved from <u>http://www.naeyc.org/ncate/files/ncate/file/faculty/Standards/NAEYC%20Initial%20an</u> <u>d%20Advanced%20Standards%203_2012.pdf</u>
- National Association for the Education of Young Children (NAEYC) & the National Association of Early Childhood Specialist in State Departments of Education (NAECS/SDE), (2003). Early Childhood Curriculum, Assessment, and Program Evaluation. Retrieved from <u>https://www.naeyc.org/files/naeyc/file/positions/pscape.pdf</u>
- National Center for Educational Statistics. (2005). National assessment of educational progress data tool. Retrieved from http://nces.ed.gov/nationsreportcard/naepdata/search.asp.

- National Research Council. (1996). *National science education standards*. Washington, DC: National Academy Press.
- National Research Council. (2007). *Taking Science to School: Learning and Teaching Science in Grades K-8.* Washington, DC: The National Academies Press. https://doi.org/10.17226/11625.
- National Research Council. (2009). *Learning Science in Informal Environments*. Washington, D.C.: National Academies Press.
- National Research Council. (2009). *Learning Science in Informal Environments: People, Places, and Pursuits. P. Bell, B. Lewenstein, & A. Schouse* (Eds.). Washington, DC.
- National Research Council. (2012). A framework for K-12 science education: Practices, crosscutting concepts, and core ideas. Committee on a Conceptual Framework for New K-12 Science Education Standards, Board on Science Education. Washington, DC: National Academies Press.
- National Research Council. (2015). *Science teachers learning: Enhancing opportunities, creating supportive contexts*. Washington, DC: National Academies Press.
- National Science Foundation. (2013). Inspiring STEM learning. (NSF Publication No. 13800). Arlington, VA: U.S. Available at: https://www.nsf.gov/about/congress/ reports/ehr_research.pdf.
- National Science Teachers Association (NSTA). (2003). NSTA Position Statement: Teacher Standards. Retrieved from <u>http://www.nsta.org/pdfs/NSTAstandards2003.pdf</u>
- National Science Teachers Association (NSTA). (2012). *Standards for science teacher preparation*. Arlington, VA: National Science Teachers Association.
- National Science Teachers Association (NSTA). (2014). NSTA Position Statement: Early Childhood Science Education. Retrieved from <u>http://www.nsta.org/about/positions/earlychildhood.aspx</u>
- National Science Teachers Association (NSTA). (2017). NSTA Position Statement: Science Teacher Preparation. <u>http://static.nsta.org/pdfs/positionstatement_teacherprep.pdf</u>
- Nayfeld, I., Fuccillo, J., & Greenfield, D.B. (2013). Executive functions in early learning: Extending the relationship between executive functions and school readiness to science. *Learning and Individual Differences, 26,* 81-88. https://doi.org/10.1016/j.lindif.2013.04.011
- Ng'asike, J. T. (2010). Turkana children's sociocultural practices of pastoralist lifestyles and science curriculum and instruction in Kenyan early childhood education (Order No. 3432417). Available from ProQuest Dissertations & Theses Global; Social Science Premium Collection. (821242754). Retrieved from

https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/82124 2754?accountid=7113

- NGSS Lead States. (2013). *Next generation science standards: For states, by states.* Washington, DC; National Academies Press.
- Nord, C. W., Lennon, J., Liu, B., & Chandler, K. (2010). *Home literacy activities and signs of children's emerging literacy, 1993, 1999, 2008*. Washington, DC: US Department of Education, Office of Educational Research and Improvement.
- Ohta, A. S. (2013). Sociocultural theory and the zone of proximal development. In Herschensohn, J.& Young-Scholten, M. (Eds.), *Cambridge handbooks in language and linguistics: The Cambridge handbook of second language acquisition.* Cambridge, UK: Cambridge University Press. Retrieved from https://libproxy.library.unt.edu/login?url=https://search.credoreference.com/content/e ntry/cuphseclaq/sociocultural_theory_and_the_zone_of_proximal_development/0?inst itutionId=4982
- Olgan, R., Alpaslan, Z. G., & Öztekin, C. (2014). Factors influencing pre-service early childhood teachers' outcome expectancy beliefs regarding science teaching. *Egitim Ve Bilim, 39*(173) Retrieved from <u>https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/15217</u> <u>19377?accountid=7113</u>
- Organisation for Economic Co-operation and Development (OECD), (2006). *Starting strong II: early childhood education and care.* Paris: OECD Publishing.
- Pendergast, E., Lieberman-Betz, R., & Vail, C. (2017). Attitudes and Beliefs of Prekindergarten Teachers Toward Teaching Science to Young Children. *Early Childhood Education Journal*, 45(1), 43–52. <u>https://doi.org/10.1007/s10643-015-0761-y</u>
- Porlan, R. & Martin del Pozo, R. (2004). The conceptions of in-service and prospective primary school teachers about the teaching and learning of science. *Journal of Science Teacher Education*, 15(1), p. 39-62.
- Preston, C. (2016). Try this: Clothes for different seasons. *Teaching Science: The Journal of the Australian Science Teachers Association*, *62*(3), 4–7. Retrieved from https://libproxy.library.unt.edu:9443/login?url=http://search.ebscohost.com/login.aspx ?direct=true&db=a9h&AN=121778871&scope=site
- Project Nexus. (2011). Drawn to science education: Studying science teaching and learning through drawings. Retrieved from <u>http://www.drawntoscience.org/researchers/scoring-rubric/index.html</u>
- Radnofsky, M. L. (1996). Qualitative models: Visually representing complex data in a image/text balance. *Qualitative Inquiry, 2(*4), 385-410.

- Reuter, J. M. (2017). Classroom preschool science learning: The learner, instructional tools, and peer-learning assignments (Order No. 10753134). Available from ProQuest Dissertations & Theses Global. (2002092043). Retrieved from <u>https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/20020</u> <u>92043?accountid=7113</u>
- Russo, S. (2006). Sharing early childhood science education in Papua New Guinea: A cross cultural experience. *Teaching Science: The Journal of The Australian Science Teachers Association*, *52*(4), 38-41. <u>https://libproxy.library.unt.edu:9443/login?url=http://search.ebscohost.com/login.aspx</u> <u>?direct=true&db=a9h&AN=24424780&scope=site</u>
- Sackes, M. (2014). Parents who want their Pre-K children to have science learning experiences are outliers. *Early Childhood Research Quarterly, 29*(2), 132-143. <u>https://doi.org/10.1016/j.ecresq.2013.11.005</u>
- Saçkes, M., Flevares, L.M., & Trundle, K.C. (2010). Four- to six-year-old children's conceptions of the mechanism of rainfall. *Early Childhood Research Quarterly, 25* (4), 536-546. doi: 10.1016/j.ecresq.2010.01.001
- Saçkes, M., Trundle, K. C., Bell, R. L., & O'Connell, A. A. (2011). The influence of early science experience in kindergarten on children's immediate and later science achievement: evidence from the early childhood longitudinal study. *Journal of Research Science Teaching*, 48, 217–235. doi:10.1002/tea.20395
- Sackes, M., Trundle, K., & Flevares, L. (2009). Using Children's Literature to Teach Standard-Based Science Concepts in Early Years. *Early Childhood Education Journal*, *36*(5), 415-422. doi:10.1007/s10643-009-0304-5
- Schreier, M. (2012). Qualitative content analysis in practice. Thousand Oaks, CA: Sage.
- Seung-Yoeun, Y. (2010). Early Childhood Teachers' Empowerment and Implementation of Teaching Method Programs for Child Development in Science Education. *Education*, 130(4), 556-560. Retrieved from <u>file:///D:/Science%20ec%20preservice/Seung-</u> Yoeun,%20Y.%20(2010).%20EARLY%20CHILDHOOD%20TEACHERS'%20EMPOWERMENT %20AND%20IMPLEMENTATION%20OF%20TEACHING%20METHOD%20PROGRAMS%20F
 OR%20CHILD%20DEVELOPMENT%20IN%20SCIENCE%20EDUCATION..pdf
- Shively, C. T., & Yerrick, R. (2014). A case for examining pre-service teacher preparation for inquiry teaching science with technology. *Research in Learning Technology, 22* doi: http://dx.doi.org/10.3402/rlt.v22.21691
- Sinclair, B. B., Szabo, S., Redmond-Sanogo, A., & Sennette, J. D. (2013). Investigating perceptions of teachers and teaching using the draw-a-teacher checklist. *Issues in Teacher Education*, 22(1), 105-123. Retrieved from

https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/15008 22398?accountid=7113

- Siry, C. A., & Lang, D. E. (2010). Creating participatory discourse for teaching and research in early childhood science. *Journal of Science Teacher Education, 21*(2), 149-160. Doi: http://dx.doi.org/10.1007/s10972-009-9162-7
- Siu, K. W. M., & Lam, M. S. (2005). Early Childhood Technology Education: A Sociocultural Perspective. *Early Childhood Education Journal, 32*(6), 353-358. <u>https://doi.org/10.1007/s10643-005-0003-9</u>
- Sleeter, C. (2010). Afterword—culturally responsive teaching: A reflection. *Journal of Praxis in Multicultural Education, 5*(1), 1-3. Published by Digital Scholarship@UNLV
- Subramaniam, K. (2007). Teachers' mindsets and the integration of computer technology. *British Journal of Educational Technology, 38*(6), 1056–1071. <u>https://doi.org/10.1111/j.1467-8535.2006.00693.x</u>
- Subramaniam, K. (2013). Student teachers' conceptions of teaching biology. *Journal of Biology Education*, 11-51.<u>https://dx.doi.org/10.1080/00219266.2013.837405</u>
- Subramaniam, K. (2013). Minority Preservice Teachers' Conceptions of Teaching Science: Sources of Science Teaching Strategies. *Research in Science Education, 43*(2), 687-709.
- Subramaniam, K. (2018). An exploratory study of student teachers' conceptions of teaching life science outdoors. *Journal of Biological Education*. First published online 16 May 2018. DOI 10.1080/00219266.2018.1472133
- Subramaniam, K., Asim, S., Lee, E. Y., & Khoo, Y. (2018). Student Teachers' Images of Science Instruction in Informal Settings: A Focus on Field Trip Pedagogy. Journal of Science Teacher Education, O(0), 1-19. <u>https://doi.org/10.1080/104656DX.2018.1452531</u>
- Tao, Y., Oliver, M., & Venville, G. (2012). Long-term outcomes of early childhood science education: insights from a cross-national comparative case study on conceptual understanding of science. *International Journal of Science and Math Education*, 10(6) 1269-1302. <u>https://doi.org/10.1007/s10763-012-9335-2</u>
- Thomas, J. A., Pedersen, J.E., and Finson, K. (2001). Validating the Draw-A-Science-Teacher-Test Checklist (DASTT-C): Exploring Mental Models and Teacher Beliefs. *Journal of Science Teacher Education*, 12(3), 295-310.
- Tippett, C.D. & Milford, T.M. (2017). Findings from a Pre-kindergarten Classroom: Making the Case for STEM in Early Childhood Education. International Journal of Science and Mathematics Education, 15(Suppl 1), 567-586. <u>https://doi.org/10.1007/s10763-017-9812-8</u>

- Trnova, E. & Trna, J. (2015). Formation of Science Concepts in Pre-school Science Education. Procedia - Social and Behavioral Sciences, 197, 2339-2346. 7th World Conference on Educational Sciences, (WCES-2015), 05-07 February 2015, Novotel Athens Convention Center, Athens, Greece. doi: 10.1016/j.sbspro.2015.07.264
- Trundle, K.C. (2009). Teaching Science during the early childhood years, Best Practice in Science Education). Retrieved from <u>http://ngl.cengage.com/assets/downloads/ngsci_pro000000028/am_trundle_teach_sc_i_early_child_scl22-0429a.pdf</u>.
- Trundle, K.C., & Sackes, M. (2015). Science and Early Education. In Pianta, R.C., Barnett, W.S., Justice, L.M., & Sheridan, S.M. (Eds.). Handbook of early childhood education. New York: A Division of Guilford.
- Trundle, K. C., & Smith, M. M. (2017). A Hearts-on, Hands-on, Minds-on Model for Preschool Science Learning. YC: Young Children, 72(1), 80-86. Retrieved from https://libproxy.library.unt.edu/login?url=https://search.proquest.com/docview/20088 19821?accountid=7113
- Tsatsaroni, A., Ravanis, K. & Falaga, A. (2005). Studying the recontextualisation of science in pre-school classrooms: Drawing on Bernstein's insights into teaching and learning practices. *International Journal of Science and Mathematics Education*, 1(4), 385-417. https://doi.org/10.1007/s10763-005-1049-2
- Tu, T. (2006). Preschool Science Environment: What Is Available in a Preschool Classroom? Early Childhood Education Journal, 33(4), doi: http://dx.doi.org/10.1007/s10643-005-0049-8
- Tudge, J. R. H., & Winterhoff, P. A. (1993). Vygotsky, Piaget, and Bandura: Perspectives on the relations between the social world and cognitive development. *Human Development*, 36, 61-81.
- Unal, M. P., Akman, B., & Gelbal, S. (2010). The adaptation of a scale for preschool teachers' attitudes towards science teaching. WCES-2010. Procedia-Social and Behavioral Science, 2, 2881-2884. Retrieved from <u>file:///D:/Science%20ec%20preservice/The%20adaptation%20of%20a%20scale%20for%</u> <u>20preschool%20teachers'%20attitudes%20towards.pdf</u>
- United Nations Educational, Scientific and Cultural Organization (UNESCO). (2006). Country profile prepared for the Education for All Global Monitoring Report 2007 Strong Foundations: Early Childhood Care and Education: Thailand Early Childhood Care and Education (ECCE) programmes. Retrieved from http://unesdoc.unesco.org/images/0014/001472/147249e.pdf

- Vandermaas-Peeler, M., Massey, K., & Kendall, A. (2016). Parent guidance of young children's scientific and mathematical reasoning in a science museum. *Early Childhood Education Journal*, 44(3), 217-224. doi: http://dx.doi.org/10.1007/s10643-015-0714-5
- Van Uum, M.S.J., Verhoeff, R.P. & Peeters, M. (2016). Inquiry-based science education: towards a pedagogical framework for primary school teachers, *International Journal of Science Education, 38*:3, 450-469, DOI: 10.1080/09500693.2016.1147660
- Vygotsky, L. S. (1987). Thinking and speech. In R.W. Rieber & A.S. Carton (Eds.), *The collected works of L.S. Vygotsky*, Volume 1: Problems of general psychology (pp. 39–285). New York: Plenum Press. (Original work published 1934.)
- Weade, R. (1987). Curriculum'n'instruction: The construction of meaning. *Theory into Practice,* 26(1), 15-25.
- Weidong, Z., & Birdsall, S. (2016). Analyzing early childhood educators' science pedagogy through the lens of a pedagogical content knowing framework. *Australasian Journal of Early Childhood*, *41*(2), 50-58.
- Westman, S., & Bergmark, U. (2014). A strengthened teaching mission in preschool: teachers' experiences, beliefs and strategies. *International Journal of Early Years Education*, 22(1), 73–88. https://doi.org/10.1080/09669760.2013.809653
- Wheatley, B. C., Gerde, H. K., & Cabell, S. Q. (2016). Integrating Early Writing into Science Instruction in Preschool. *Reading Teacher*, *70*(1), 83-92. doi:10.1002/trtr.1470.
- Windschitl, M. (2003). Inquiry projects in science teacher education: What can investigative experiences reveal about teacher thinking and eventual classroom practice? *Science Teacher Education*, *87*, 112-143.
- Yoo, S. (2009). Using Portfolios as a learning tool to develop preservice teachers' inquiries and perspectives in early science teaching in south Korea. *Education*, *130*(1), 72-77. Retrieved from
 <u>file:///D:/Science%20ec%20preservice/Yoo,%20S.%20(2009).%20USING%20PORTFOLIO</u>
 <u>S%20AS%20A%20LEARNING%20TOOL%20TO%20DEVELOP%20PRESERVICE%20TEACHER</u>
 <u>S'%20INQUIRIES%20AND%20PERSPECTIVES%20IN%20EARLY%20SCIENCE%20TEACHING</u>
 %20IN%20SOUTH%20KOREA..pdf
- Yoon, J., & Onchwari, J. A. (2006). Teaching young children science: Three key points. *Early Childhood Education Journal, 33*(6), 419-423. doi: http://dx.doi.org/10.1007/s10643-006-0064-4
- Zhai, J., Jocz, J.A., & Tan, A. (2014). 'Am I Like a Scientist?': Primary children's images of doing science in school. *International Journal of Science Education*, *36* (4), 553–576. http://dx.doi.org/10.1080/09500693.2013.791958

- Zhang, W. & Birdsall, S. (2016). Analysing early childhood educators' science pedagogy through the lens of a pedagogical content knowing framework. *Australasian Journal of Early Childhood, 41*(2), 50-58. Retrieved from file:///D:/Science%20ec%20preservice/Weidong,%20Z.,%20&%20Birdsall,%20S.%20(201 6).%20Analysing%20early%20childhood%20educators'%20science%20pedagogy%20thr ough%20the%20lens%20of%20a%20pedagogical%20content%20knowing%20framewor k..pdf
- Zucker, T. A., Williams, J. M., Bell, E.R., Assel, M. A., Landry, S. H., Monsegue-Bailey, P., Crawford, A., & Bhavsar, V. (2016). Validation of a brief, screening measure of lowincome pre-kindergarteners' science and engineering knowledge. *Early Childhood Research Quarterly, 36*, 3rd Quarter, 345 – 357. <u>https://doi.org/10.1016/j.ecresq.2015.12.018</u>